Ilja A. Seržant*

Typology of partitives

https://doi.org/10.1515/ling-2020-0251
Received November 17, 2020; accepted November 17, 2020; published online

Abstract: This paper explores the coding patterns of partitives and their functional extensions, based on a convenience sample of 138 languages from 46 families from all macroareas. Partitives are defined as constructions that may express the proportional relation of a subset to a superset (the true-partitive relation). First, it is demonstrated that, crosslinguistically, partitive constructions vary as to their syntactic properties and morphological marking. Syntactically, there is a cline from loose - possibly less grammaticalized - structures to partitives with rigid head-dependent relations and, finally, to morphologically integrated one-word partitives. Furthermore, partitives may be encoded NP-internally (mostly via an adposition) or pronominally. Morphologically, partitives primarily involve markers syncretic with separative, locative or possessive meanings. Finally, a number of languages employ no partitive marker at all. Secondly, these different strategies are not evenly distributed in the globe, with, for example, Eurasia being biased for the separative strategy. Thirdly, on the functional side, partitives may have functions in the following domains in addition to the true-partitive relation: plain quantification (pseudo-partitives), hypothetical events, predicate negation and aspectuality. I claim that the ability to encode plain quantification is the prerequisite for the other domains. Finally, it is argued that there is a universal preference towards syncretism of two semantically distinct concepts: the proportional, true-partitive relation (some of the books) and plain quantification (some books).

Keywords: aspectuality; negation; partitives; pseudo-partitives; typology; quantification

1 Introduction

The aim of this paper is to explore the crosslinguistic variation of the coding patterns of partitive constructions and functional extensions that partitive constructions are found to attest in different languages.

^{*}Corresponding author: Ilja Seržant, Department of Slavonic Studies, Christian-Albrechts-Universität zu Kiel, Leibnizstraße 10, D-24098 Kiel, Germany, E-mail: ilja.serzant@slav.uni-kiel.de

Crosslinguistically, partitive constructions vary as to the degree of the internal integration, syntactic type and marking strategy. Partitive constructions may be formed by very diverse markers that range from adpositions and case affixes to verbal clitic particles or partitive pronouns (cf. Budd 2014; Hoeksema 1996b; Koptjevskaja-Tamm 2009, 2001; Luraghi and Kittilä 2014; Tănase-Dogaru 2017) or they may involve no marker at all.

When it comes to functional extensions, only the development of partitives into co-expression markers of predicate negation has received sufficient attention in the typological literature (Arkadiev and Kozhanov, forthcoming; Budd 2014; Miestamo 2014). By contrast, interactions of partitives with grammatical domains such as hypothetical predication, predicate quantification and aspectuality – devoted much attention in formal semantics (e.g., on aspect, cf., *inter alia*, Filip 1999 (especially Ch. 6); Kiparsky 1998; Krifka 1992) – have not been discussed in the typological literature (except Budd 2014). For example, the partitive case in Finnic languages (primarily Finnish) is famous for its various functional facets. In addition to the original and, by now, rare function of encoding the part-whole relation (1a), it has acquired new meanings of indefinite quantity (1a) as well as various clause-level functions such as the co-expression of predicate negation (2), aspectuality, cf. (3a) as opposed to (3b), or even hypothetical events, cf. (4a) (cf. Larsson 1983: 84; de Hoop 2003 among many others):

(1) Finnish (Uralic)

Sö-i-n puuro-a
eat-PST-1SG porridge-PART.SG
a. 'I ate some of the porridge.'
b. 'I ate some porridge.'
(Huumo 2010: 91)

(2) Finnish (Uralic)

En voidel-lut suks-i-a.

NEG.1SG wax-PST.CONNEG ski-PL-PART

'I did not wax (the) skis.'

(Huumo 2010: 95)

(3) Finnish (Uralic)

a. Lyhens-i-n hame-tta
 shorten-PST-1SG skirt-PART.SG
 'I was shortening the skirt.' or 'I shortened the skirt [somewhat].'

CORRECTED PROOF

DE GRUYTER MOUTON

Typology of partitives — 3

b. Lyhens-i-n hamee-n
 shorten-PST-1SG skirt-ACC.SG¹
 'I made the skirt short.'
 (Huumo 2010: 95)

(4) Finnish (Uralic)

a. *Luul-i-n hän-tä viisaaksi*. think-PST-1SG 3SG-PART wise

'I (mistakenly) thought him to be wise.' (but he turned out not to be)

b. Luul-i-n häne-t viisaaksi. think-PST-1SG 3SG-ACC wise 'I thought him to be wise.' (Huumo, p.c.)

In this paper, I proceed as follows. Section 2 sets out the main concepts underlying this study and serves, thereby, as a further introduction to partitives with a summary in Section 2.3. I start with purely semantic concepts of true-partitivity and plain quantification in Section 2.1 and then turn to comparative concepts that are based on both encoding and function in Section 2.2. Here, I introduce and define the following terms: partitive constructions (alias partitives), quantifier, restrictor in Section 2.2.1, implicit expressions (Section 2.2.2), generalized partitives (Section 2.2.3) and pseudo-partitives (Section 2.2.4). Section 3 provides information about the sample and the database. Section 4 describes the constructional variation of partitives: variation in the degree of internal integration of partitive constructions (Section 4.1) and the coding variation (Section 4.2), summarized in Section 4.3. Section 5 is devoted to the areal distribution of marking strategies. Section 6 discusses functional extensions that partitives sometimes have in different languages: pseudo-partitives and plain quantification (Section 6.1), partitives (co-) expressing hypothetical events (Section 6.2), partitives co-expressing predicate negation (Section 6.3) and partitives conditioning event quantification and thus contributing to the aspectual interpretation of the predicate (Section 6.4). Section 6.5 describes the frequency distributions of these functional facets and draws generalizations. Section 7 provides argumentation for a universal trend toward the syncretism of the true-partitive relation and plain quantification. Finally, Section 8 summarizes the results and presents conclusions.

¹ I take the Finnish genitive in the singular and nominative in the plural as a syncretic accusative case, as the personal pronouns – the only ones to distinguish a dedicated accusative case – show the accusative case here.

2 Main concepts: true-partitive relation, implicit expressions, pseudo-partitives, generalized partitives

Unfortunately, the term *partitive* has been used in a broad variety of meanings in the literature, comprising the purely semantic concept of the part-whole relation but also particular markers capable of encoding very diverse functions such as the partitive case in Finnish or the partitive article in Romance. Sometimes "partitive" is even used for any kind of expression in a given language, where another language would have a partitive marker. While it is impossible to adopt all traditional uses, in what follows, I present the terminological framework adopted in this paper which tries to accommodate some of the traditional uses to the extent reasonable for this study. Thus, my definition of partitives, given in (9), is very broad in order to maintain continuity with most of the previous work. It is essential to provide general definitions that are not based on language-specific properties in order to allow for typological comparison (cf. Haspelmath 2018).

I first define the semantic concepts of the *true-partitive relation*, subset and superset (Section 2.1) and then introduce constructional concepts in Section 2.2. Section 2.3 summarizes the main concepts.

2.1 Semantic concepts: true-partitive relation, subset, superset and plain quantification

Consider the following example:

(5) Lithuanian (Indo-European)

Vien-as jo kolegų pasakojo, kad...

one-NOM.SG 3SG.M.GEN colleague.GEN.PL tell.PST.3 that

'One of his colleagues said that...'

In (5), the subject participant 'one' is a one-member *subset* of the well-defined *superset* 'his colleagues' – a relation that is coded by the genitive case on the superset NP 'colleagues'. The remainder (*the complement*) contains those 'of his colleagues' who have not participated in the event in (5) (cf., among others, Enç 1991; Koptjevskaja-Tamm 2001, 2009; von Heusinger 2002: 261–62 for these terms). The subset and the superset relate to each other as a part to its whole. This relation renders the concept of a proportion. The same relation may also be found with mass nouns, cf. (6):

(6) a cup of the milk in the fridge

In (6), it is more appropriate to speak about portions instead of sets. However, for the sake of brevity I will only speak about subsets and supersets, subsuming both divisible and non-divisible types of referents. In what follows, I refer to the relation thus defined as to the *true-partitive relation* ("real partitivity" in Selkirk 1977):

(7) The true-partitive relation

A true-partitive relation obtains when there is a subset-superset relationship between two sets (with mass nouns: two portions) of the same kind.

Thus, in contrast to *plain quantification* found in the English expression *one colleague* or in *some students*, the true-partitive relation involves proportional quantification, i.e., there is a proportion between the quantity of the subset and the quantity of the superset.

It follows from the definition that in order for a linguistic expression to have the true-partitive meaning, the superset must have a definite specific interpretation (non-generic and non-property-denoting/predicate, non-distributive, and discursively accessible.)² This requirement has been referred to as *Partitivity Constraint* in the literature (first observed in Jackendoff 1977; later analyzed and refined in Barwise and Cooper 1981; de Hoop 1997, 2003: 186 following Dowty and Brodie 1984; Hoeksema 1996a: 6; Ionin et al. 2006; Ladusaw 1982; Reed 1989; Tănase-Dogaru 2017; Westerståhl 1985).³ Accordingly, while (6) is truly partitive, (8) – even though being formally very similar to (6) – is not:

(8) a cup of tea

This is because (8) violates the Partitivity Constraint in that *tea* is indefinite-generic here.

Furthermore, the definition in (7) comprises only proportional part-whole relations, i.e., those relations in which both the subset and the superset consist of objects of the same kind. For example in (5), the subset 'one colleague of his' belongs to the same kind of objects ('his colleagues') as the superset ('his colleagues'). By contrast, there are other, non-proportional types of the part-whole relation such as *chicken-feet*, *church-door* or *mountain-top* (Hoeksema 1996a: 1) in which the part (*the chicken's feet*) does not consist of objects of the same kind as the

² Distributive refers to distributional quantifiers such as *each* in English.

³ There are exceptions to this which have been widely discussed in the semantics literature on partitives, e.g. *that book could belong to one of three people*, where *three people* is indefinite (de Hoop 2003: 183), and various attempts have been made to provide an account for these (cf. de Hoop 1997; Ladusaw 1982). Moreover, certain definite NPs are nevertheless excluded from occurring in the restrictor, such as those headed by *both* or in many instances *all*.

whole (*the chicken*). I refer to these latter relations as to *meronymics* to clearly distinguish them from partitives.⁴

2.2 Constructional concepts

In this section, I define constructional concepts of this study that are defined in terms of both the pattern of encoding and its function: *partitives* (Section 2.2.1), *implicit expressions* (Section 2.2.2), *generalized partitives* (Section 2.2.3) and *pseudo-partitives* (Section 2.2.4). These concepts are comparative concepts (as per Haspelmath 2018).

2.2.1 Partitives, quantifier and restrictor

Languages have different means of encoding the true-partitive relation. Grammatical means that are capable of encoding the true-partitive relation are referred to as *partitive constructions* or simply as *partitives* in this paper:

(9) Definition of partitives

Partitives are grammatical constructions that may be used to encode the truepartitive relation without relying on contextual inferences. Partitives
obligatory encode (i) a quantifier and (ii) the restrictor. Partitives are often

encoded by (iii) a special partitive marker or lexically.

Contextual inferences are understood in the narrow sense, excluding anaphora resolution. Thus, partitive pronouns (such as French *en* 'thereof') *encode* – and do not implicate – the reference to the superset in the same way as, for example, a third-person pronoun is a reduced form of encoding the reference. Moreover, many pronouns may also be used deictically and, on this reading, they should be able to occur in out-of-the-blue contexts with the true-partitive meaning.

In a true-partitive relation (7), the *quantifier* corresponds to the subset and the *restrictor* to the superset. While subset and superset are terms confined to the true-partitive relation, the *quantifier* and the *restrictor* will be used more generally in this paper for any kind of partitive constructions, including pseudo-partitives (see Section 2.2.4 below).

Crucially, the partitive construction must be available – even if restrictedly – for encoding the true-partitive relation in the language in order to count as a partitive at all. This is also the original use of the term. While this definition covers most of the expressions that have been referred to as partitives in the literature,

⁴ I thank one of the anonymous reviewers for suggesting this term.

there are some exceptions. For example, I have not come across any examples of the partitive case in -(r)ik in Basque being used for encoding the true-partitive relation (e.g. in Hualde and Oritz de Urbina 2003, eds.: 124–126) while only the ablative case in Basque (-etatik) has the ability to encode the true-partitive relation. Accordingly, the case in -(r)ik – traditionally referred to as partitive – is not a partitive in the framework of this paper, whereas the Basque ablative case is.

Having said that, the definition in (9) leaves much space for other aspects of partitives. First, this definition allows partitives to have other meanings in addition to the true-partitive relation (see Section 6). As I argue below (Sections 6–7), languages rarely have dedicated grammatical means that are only capable of encoding the true-partitive relation. For example, the Finnish partitive is capable of expressing many more meanings than just the true-partitive relation. In fact, the latter is even much rarer than the other meanings of the partitive case. Thus, the following corollary to the definition in (9) can be formulated:

(10) A corollary to the definition in (9)

In addition to encoding the true-partitive relation, partitive constructions may also have other (diachronically) related functions.

2.2.2 Implicit expressions

The definition in (9) excludes *implicit expression of a true-partitive relation* (*implicit partitives* in von Heusinger and Kornfilt 2017; or *covert partitives* in de Hoop 2003: 207), i.e., quantifiers, numerals and other expressions that may occur in a context in which the partitive interpretation is likely (henceforth *implicit expressions*). This is found when the superset may be implicated. For example, in (11), the quantifier phrases *some/three/a few flowers* do not contain the reference to the superset, and the superset can only be inferred given the first clause as *the flowers in the garden*:

(11) There are flowers in the garden. Bring me **some/three/a few flowers**.

Therefore, it can be said that implicit expressions do not themselves encode the true-partitive relation. For example, they are unlikely to be interpreted as such in out-of-the-blue contexts with no presuppositional support. Moreover, it would be counter-intuitive to say that *three* is a grammatical expression to encode the true-partitive relation; nor do *some* or *a few* encode the true-partitive relation in English.⁵ Rather, these simply quantify items of a particular kind (henceforth *plain quantification*). While I acknowledge the tradition of subsuming these under

⁵ Alternatively, *some* may be considered as being ambiguous by having two allomorphs in English: the stressed partitive allomorph and the unstressed non-partitive one (cf. *stressed indefinites* in Hoeksema 1996a: 2).

partitives as well (*inter alia*, Budd 2014; Crowley 1982; Hoeksema 1996a, b; Mosel and Hovdhaugen 1992; Schneider 2010),⁶ I refrain from doing so here in order to have more fine-grained terminological distinctions.

Thus, a typical property of implicit expressions is that they do not encode the reference to the superset (*the flowers in the garden* in (11)) even not in a reduced or a lexicalized form, but only an implicational reference to the kind (*flowers*), thus violating the definition in (9).

In contrast to English *some*, e.g., German *davon* 'thereof' is a partitive:

- (12) German (Indo-European)
 - a. Es gibt Blumen im Garten.'There are flowers in the garden.'
 - b. *Bring mir fünf davon*. bring me five thereof 'Bring me five of them.'

German *davon* 'thereof' does encode the superset which is resolved either anaphorically, as in (12a-b), or deictically (if (12b) is uttered out of the blue by pointing with a finger to some items). Having said this, indefinite pronouns such as English *some* may come functionally very close to a partitive. For example, the German pronoun *welche* 'some' (homonymous with the interrogative) may indeed be considered to be a partitive as it patterns very much like *davon* (Glaser 1992).

To summarize: In order to analyze an expression as a partitive and not as an implicit expression, two conditions have to be met: (i) there must be an example in which this expression encodes the true-partitive relation without contextual inferences and (ii) the reference to the superset must be encoded in this example, i.e., be retrievable deictically, anaphorically or by a full NP.

In contrast to English, some languages do have dedicated indefinite quantifiers that are partitives lexically because they can only be used in proportional quantification but not in plain quantification. For example, Cora (Uto-Aztecan; Mexico) has dedicated indefinite quantifiers to encode the true-partitive relation, cf. *héiwa* 'many (non-partitive)' vs. *mwi'iká-ka* 'many-ACC (partitive)' (Casad 1984: 265); Haida (Haida) has dedicated, lexical partitive quantifiers such as *t'iij* 'some of' (Enrico 2003: 771 and elsewhere).⁷

⁶ According to Budd (2014: 535) there is a distinction between *he* 'some' and (*t*)ē*tahi* 'a/some' followed by the genitive preposition in Maori, in which the latter expresses the true-partitive meaning 'some of'. However, Bauer (1993: 297) provides a series of examples of the non-partitive use of *ētahi* in his grammar as well. So, *ētahi* might not be a dedicated partitive quantifier in Maori.

2.2.3 Generalized partitives

Partitives encoding the true-partitive relation are often binominal, i.e., consisting of two nominals (i.e. nouns, pronouns, pronominal numerals, etc.): the quantifier and the restrictor nominal. However, there are considerable input restrictions on both. The restrictor of a true-partitive is always definite (Partitivity Constraint). The quantifier typically does not admit adjectival quantifiers like every, definite and indefinite articles. Personal pronouns are also typically excluded from occurring in the quantifier position (de Hoop 2003: 180 with references). More generally, there is a very strong tendency for the quantifier to be indefinite. Thus, many grammars consulted in this study do not even provide examples of partitives with definite quantifiers (except for superlative constructions relying on partitive constructions). What is more, not only is the quantifier most frequently indefinite but it is also most frequently pronominal (i.e., lacking a full noun). The quantifier position is filled predominantly with one, some, none (e.g., some of them), much, many (e.g., much of this stuff) or an indefinite numeral (e.g., five of the students) and very rarely by an indefinite full NP.9 The latter is a very rare situation in which the reference to the ontological kind is given in the quantifier NP:

(13) Hinuq (Nakh-Daghestanian)

[tor-tes sadaq-za-tes] q wena uži: rux-iš mašina-be
three-ABL all-OBL.PL-ABL two.OBL boy.ERG buy.PST car-PL
'Two boys out of three / all bought cars.'

(Forker 2013:722)

Indeed, a small corpus survey of the Russian partitive *iz nix* [lit.] 'from them' in the oral subcorpus of the Russian National Corpus illustrates this strong bias towards indefiniteness and pronominalization of the quantifier (Table 1). I have analyzed the first 300 hits of this expression. After having excluded all nonpartitive instances such as predicates governing the preposition *iz* 'from', 277

⁷ It may even be said that the English stressed allomorph of some – often referred to as SOME as opposed to the unstressed sm in the literature – is a dedicated partitive expression (a partitive pronoun) as well if it can be shown that it encodes the true-partitive relation.

⁸ In split constructions, a focal personal pronoun may appear though, e.g., in Russian: *Iz posetitelej tol'ko ja byl pri den'gax* [lit. from visitors only I was with money] 'Among the visitors, it was only me who had money.'

⁹ The only seeming exceptions are various nominal quantifiers and measure phrases such as *a cup of, a bottle of, a group of* which, however, demonstrably do not behave in the same way as full nouns in many respects. Strong quantifiers such as *all, both, none* do not occur in the partitive construction in a number of languages, e.g., in Malagasy (Paul 2000: 125).

Table 1: The relative frequency of the definite vs. indefinite subsets to the superset *iz nix* 'from/of them' in the Russian National Corpus (www.ruscorpora.ru), the Spoken subcorpus.

	Indefinite					Definite	
	quantifiers ¹⁰	indefinite pronouns ¹¹	interrog. ¹²	numeral	'none'13	superlative	other
	115 (44%)	80	28	21	18	11	4
Total	262 (95 %)					15 (5 %	6)

instances of the true-partitive relation were left of which ca. 95% had indefinite quantifiers and 0% quantifiers with a full NP.

This frequency disproportion makes sense from the semantic point of view: a definite, full NP contains exhaustive referential information and its referent is immediately identifiable. Any additional referential information such as the reference to the superset is, therefore, superfluous (cf. ***these students of my students*). ¹⁴ By contrast, an indefinite referent often needs additional referential support for discourse cohesion. Moreover, pronominal input is also motivated because the superset already contains the information on the kind and resuming it in the subset would not be efficient (cf. ***some students of my students*).

Crucially, these frequency asymmetries have effects on the grammar of many partitives. Thus, the strong affinity to indefinite, pronominal referents is the main motivation partitives which conventionally leave the quantifier unexpressed, cf. 'some of his colleagues' in (14a) vs. '[some] of his colleagues' in (14b).

- (14) Lithuanian (Indo-European; p. k.)
 - a. *Mačiau keletą jo kolegų*.
 see.PST.1SG some.ACC 3SG.GEN colleague.GEN/PART.Pl
 'I saw some of his colleagues.'
 - b. *Mačiau jo kolegų.* see.PST.1SG 3SG.GEN colleague.GEN/PART.Pl 'I saw some of his colleagues.'

The frequency asymmetries create the default expectation of the quantifier being an indefinite quantifier used pronominally, which, therefore, need not be expressed. Many languages conventionalize partitives with the quantifier left unexpressed and understood as 'some/any'. I refer to this mono-nominal subtype

¹⁰ Includes: nekotorye 'some', kakie-to 'some', neskol'ko 'some', každyj 'each', mnogie 'many'.

¹¹ Includes: odin 'one', ljuboj 'any', kakoj-to 'any', kto-to 'a person', drugie 'others'.

¹² Includes: skol'ko 'how many', kotoryj 'which', kto 'who', kogo 'whom'.

¹³ Includes: nikto 'no one', ni odin 'not any one'.

¹⁴ Again, the only exception is superlatives, which feature definite subsets by definition.

of partitives as to *generalized partitives* (other terms used in the literature are *bare partitives* in Hoeksema 1996a: 15; or *independent partitives* in Seržant 2014a, 2014b, 2015a, 2015b). The development of generalized partitives is a very frequent phenomenon found with partitives: 39% (47/122) of all marked partitives in my database allow for leaving the quantifier unexpressed but understood as 'some/any'. ¹⁶

The conventionalized indefinite quantifier 'some/any' entails a number of properties of generalized partitives crosslinguistically. First, generalized partitives typically encode low-individuated referents and, accordingly, tend to be formed out of NPs that themselves typically occur as low-individuated, e.g., mass nouns (such as 'water') or plurals (e.g. 'flowers'). As a consequence, generalized partitives tend to encode referents that are episodic and have a low probability of recurring in the following discourse (Helasvuo 1996 for Finnish; Tveite 2004: 150 for Livonian; Seržant 2012 for Ancient Greek; Seržant 2014a: 267 for Lithuanian; Seržant 2014b: 307–309 for North Russian).

Secondly, another consequence of low individuation is that generalized partitives primarily co-occur with those lexical verbs that cross-linguistically tend to demote or leave out their objects most frequently (in terms of type and token frequency), e.g., via antipassives. These are verbs such as 'eat', 'drink', 'hear' or 'read' (Næss 2007: 127; Seržant et al. forthc). These verbs typically constitute the core of those verbs that allow for partitive objects. Thus, if partitives are not very productive in a language, then it is these verbs that are most likely to be found with partitives. For example, the partitive use of the ablative case in Mordvin languages (the separative strategy) is not very productive (if compared to the related Finnish, for example). It is these verbs that are found with the partitive ablative objects in Mordvin but not with verbs that do not tend to demote their objects, such as 'kill' or 'open'. In other languages, the latter are primarily found with partitives that have predicate-level functions (Sections 6.2–6.4).

Generalized partitives, once conventionalized, may develop into a category in its own right, often having functional facets that are not found with partitives with an explicit quantifier. For example, generalized partitives may compete with other strategies of argument marking, yielding a special subtype of differential-argument-marking systems (cf. Witzlack-Makarevich and Seržant 2018: 15–16). Notably, in this role, generalized partitives fail to differentiate between A, S and P and mark all three roles alike. For example, the partitive prefix/clitic ?a'-/aa- in Tlingit (Athabaskan-Eyak-Tlingit) can replace subject and object prefixes (Leer 1991: 123–124; though see Crippen 2012: 327), and similarly the partitive

¹⁵ Note that indefinite referents are frequently left unexpressed cross-linguistically, which is a phenomenon *sui generis* not confined to partitives only. However, once quantifier-less partitives (i. e., *generalized partitives* below) have developed, they may undergo pathways that are not found with indefinite pronouns.

¹⁶ Unmarked partitives (Section 4.2.2.5) are excluded.

pronouns in Eibela (Bosavi) do not distinguish between all three roles. Armenian (Indo-European) may have partitive-ablative intransitive subjects and transitive objects, thus S=P (Dum-Tragut 2009: 313). Furthermore, a number of Finnic varieties occasionally allow all three roles to be coded by the partitive case, i.e., some A's and S's and nearly all P's (Huumo 2018; Koptjevskaja-Tamm and Wälchli 2001: 658; Lytkin et al. 1975: 108; Seržant 2015a: 396). Accordingly, the following cline of preference obtains from the least likely on the left:¹⁷

(15) A < S < P

It is the inherently indefinite meaning of generalized partitives that reduces compatibility of generalized partitives with A and partly with S, since the former but sometimes also the latter have a strong preference for definite inputs (cf. Comrie 1981: 128; Croft 1988; Dahl 2000: 50). Moreover, generalized partitives have a considerable overall effect on the semantic transitivity of the hosting clause (cf. Næss 2004, 2007). To this end, partitives in the P slot come functionally – and sometimes even morphologically (when incorporating a partitive pronoun or a quantifier) – very close to antipassives (cf. e.g., the partitive-like meaning facets of the antipassive in Sochiapan (Otomanguean; Mexico) in Foris 2000: 285). In some languages, the effect on transitivity is even rendered morphologically. For example, in Apma, the partitive marker -te – used in (16b) as a pseudo-partitive (Section 2.2.4) – blocks the transitivity marker -e that is otherwise obligatory in this language, cf. (16a):

- (16) Apma (Austronesian)
 - a. Te gel-e val?

 3sg.pfv buy-tr house
 'Did he buy the house?'
 - b. Te gel-(*e)=te val?
 3SG.PFV buy-*TR-PART house
 'Did he buy any/some house?'
 (Schneider 2010: 164)

To summarize, generalized partitives are a subtype of partitives that may exhibit properties that are not found with the corresponding partitives with an explicit quantifier.

¹⁷ Quite rarely, generalized partitives may also occur in other syntactic positions as well. For example, Ancient Greek allows generalized partitives to occur in syntactic positions that are otherwise occupied by non-structural accusatives or datives (Seržant 2012: 196–197).

2.2.4 Pseudo-partitives

While the semantic concept of the true-partitive relation necessarily involves two distinct sets or portions, partitives may sometimes also express quantification of just one single set or a portion:

- (17) Yesterday I had a cup of tea.
- (18) Yesterday I had a cup of the tea that you made for me.
- (17) is a construction that morphologically but not necessarily syntactically 18 coincides with a construction that codes the true-partitive relation in English as in (18). However, no true-partitive relation obtains in (17) because there is no clearly defined superset in the restrictor position. Instead, one finds a kind-referring expression tea, which only provides the reference to the ontological kind of substance in the cup but not to a particular referent.

With a kind-referring expression, neither the complement nor the superset can be defined in terms of sets or portions. Accordingly, (17) is semantically a plain measure phrase. However, it is formally somewhat more complex than a plain measure phrase like *much tea* because it involves the partitive marker *of*. Measure phrases such as (17) have been referred to as *pseudo-partitives* following Selkirk (1977), e.g., in Koptjevskaja-Tamm (2001, 2009).

(19) Definition of pseudo-partitive constructions

A pseudo-partitive construction (abbreviated: a pseudo-partitive) is a partitive construction with no specific superset in the restrictor.

While the true-partitive relation implies proportional quantification, pseudo-partitives denote plain quantification such as amounts (*a group of people*), measures (*a cup of tea*) or quantities (*a lot of people*, *a majority of people*) of particular kinds (*people*, *tea*). Therefore, pseudo-partitives are sometimes referred to as *quantitative partitives* (e.g., Ihsane 2013). Accordingly, pseudo-partitives do not encode a relation between two referents but rather encode just one referent that is quantified or measured. Semantically pseudo-partitives pattern with simple measure or quantifier phrases such as *many people* in English or *eine Gruppe Touristen* (lit. 'a group tourists') 'a group of tourists' in German and tend to reduce

¹⁸ Selkirk (1977) puts forward syntactic tests which show that there is also a difference in the syntactic properties between true partitives and pseudo-partitives, such as the possibility of extraction of the head NP with partitives but not with pseudo-partitives.

¹⁹ Pseudo-partitives may be subdivided into further subclasses (Koptjevskaja-Tamm 2001).

their original syntactic structure of one NP embedded into another NP towards just one NP (see Selkirk 1977 on English; de Hoop 2003 on Dutch).

In what follows, I make two amendments to the traditional use of the notion pseudo-partitive in the literature. First (i), I argue against treating it as a purely semantic notion with no regard to structural criteria. Second (ii), I argue that pseudo-partitives, like partitives expressing the true-partitive relation, may also occur not only with an explicit quantifier but also without the quantifier, i.e., as *generalized pseudo-partitives*.

As for (i), there is a tradition to extend the notion of pseudo-partitives to include any kind of measure phrases, including those that have nothing to do with partitives in the language in question. For example, *Glas Wein* (lit. 'glass wine') 'glass of wine' in German is a pseudo-partitive according to some researchers (e.g., de Hoop 2003: 192; Koptjevskaja-Tamm 2009, 2001; Tănase-Dogaru 2017) while it is not a pseudo-partitive in the terminology adopted in this paper. The reason is that such an extension of a structural-semantic term into a purely semantic notion unnecessarily overgeneralizes the original term of Selkirk (1977), making it synonymous with the more transparent term *measure phrase*, or *quantifier phrase* for that matter. Moreover, this conceptual extension also produces confusion in that a *glass of wine* in English can no longer be terminologically distinguished from *Glas Wein* in German despite the fact that these two involve morphologically and syntactically different sorts of constructions.

Selkirk (1977) made the important discovery that a *glass of wine* in English – although morphologically similar to the expression of the true-partitive relation in English (with its head and dependent NPs) – patterns syntactically (and semantically) like one NP in many respects. It is in this sense that the term *pseudo-partitive* is justified for the English expression *a glass of wine*. The latter is indeed a seeming – hence "pseudo" – partitive. By contrast, the German *Glas Wein* is straightforwardly analyzable as one NP and has no structural, semantic or morphological affinity to true-partitives in German whatsoever. The latter are encoded by means of the preposition *von* 'from' (e.g., *viele von ihnen* 'many of them') or with the genitive case (e.g. *einer meiner Studenten* 'one of my (GEN) students'). Its only relation to partitives is based on the fact that *Glas Wein* is a measure phrase that may be translated by a partitive-like expression in some other languages including English.

Another problem is that if pseudo-partitive is taken as a purely semantic notion, then one might misleadingly conclude that, for example, *much wine* in English is a pseudo-partitive because it is translated with a partitive-like construction into Russian (with the genitive originally stemming from the possessive partitive strategy) or Basque (with the partitive case) and because it is synonymous with the pseudo-partitive *a lot of wine* in English. This purely semantic use of the term *pseudo-partitive* is also unfortunate for the description of the diachronic process by which true-partitives first develop into pseudo-partitives and only after

that into plain quantifier phrases (Seržant, submitted). I conclude that the purely semantic use of the term *pseudo-partitives* is ill-advised and, instead, structural affinity to partitives must be part of the definition of pseudo-partitives (19):²⁰

(20) A corollary of definitions (9) and (19)

Pseudo-partitives are only found if they exploit the grammatical means that, at the same time, may also be used to encode the true-partitive relation in the language.

As for (ii), another important aspect of pseudo-partitives is that there are also generalized pseudo-partitives, i.e., pseudo-partitives that occur without an explicit head quantifier. Traditionally, only pseudo-partitives with an explicit, "nouny" quantifier or a measure phrase such as a group of, a bunch of, a cup of, a bottle of, etc. were considered pseudo-partitives (inter alia, Koptjevskaja-Tamm 2001, 2009). However, similar to partitives with explicit quantifiers (21a), pseudo-partitives may also occur as generalized partitives (21b), the only difference being the presence (21a) vs. absence (21b) of the quantifier:

- (21) Russian (Indo-European)
 - a. Voz'mi nemnogo / gorst' konfet take.IPV some / handful candy.GEN(=PART).PL 'Take some / a handful of candies!'
 - b. Voz'mi konfet take.IPV candy.GEN(=PART).PL 'Take [some] candies!'

Crucially, neither (21a) nor (21b) represents an instance of the true-partitive relation because the restrictor NP 'candies' is indefinite generic; contrast (21) with (22):

- (22) Russian (Indo-European)
 - a. Voz'mi nemnogo von tex konfet take.IPV some DEICTIC DEM.GEN(=PART).PL candy.GEN(=PART).PL 'Take some of those candies there!'
 - b. Voz'mi von tex konfet
 take.IPV DEICTIC DEM.GEN(=PART).PL candy.GEN(=PART).PL
 'Take some of those candies there!'

I conclude that both (21a) and (21b) are pseudo-partitives according to the definition in (19).

²⁰ Instead, the more traditional term *measure phrase* should be used for the constructions of the type *a glass of wine* or *a packet of eggs*. Then, one can say that some measure phrases are coded by pseudopartitive constructions (e.g., in English *glass of wine*) while other measure phrases resort to the construction that is generally used for plain quantification in the language (e.g., in German *Glas Wein*).

2.3 Summarizing the main concepts

I summarize the main concepts of this paper. The definitional property of partitives is their ability to encode the true-partitive relation (9), which does not mean that partitives always have to express the true-partitive relation only (10). In fact, such a narrow type of partitives is not so frequent crosslinguistically. Furthermore, I have argued that the distinction between explicit-quantifier partitives vs. generalized partitives is orthogonal to the distinction between the true-partitive vs. pseudopartitive usage and, in principle, all four combinations are possible: Examples (21a) and (22b) are partitives with an explicit quantifier. The partitive in (22b) encodes the true-partitive relation, the partitive in (21a) is a pseudo-partitive (encoding plain quantification). In parallel, examples (21a) and (22b) are *generalized partitives*. I summarize this in Table 2.

Table 2: Ontology of partitives.

	Partitives			
	Encoding the true-partitive relation	Pseudo-partitive		
head quantifier generalized partitive	the quantifier and the restrictor are encoded only the restrictor is encoded			

3 The data

The discussion in the remainder of this paper is based on a database consisting of 171 partitives. Each partitive has been tagged along the following variables: *language* of attestation, *language family*, macroarea, syntactic *type*, marking *strategy*, *dedicated* partitive and the ability to pattern as a *pseudo-partitive*.

The variable *syntactic type* contains the basic syntactic information about the locus of the partitive marking and may have the following two values: *NP-internal* encoding (adpositions and possessive indexes) and *pronominal* encoding (partitive pronouns or quantifiers).

The variable marking *strategy* contains the information about the morphological marker of each partitive and may have the following values: *possessive*, *separative*, *unmarked*, *locative* and *unclear*. Note that, strictly speaking, the value *unmarked* is both a marking strategy (zero morphological marking) and a syntactic type (juxtaposition).

Finally, whether a partitive marking is *dedicated* (i.e., signifying partitive functions only) or whether a partitive may also pattern as a *pseudo-partitive* has been tagged with just *yes* or *no*.

The database (published in Seržant 2020) was created on the basis of a convenience sample of partitive expressions covering 138 languages (Figure 1). The sample is biased toward languages of Eurasia (48% of the entries, 82/171). The data were collected from grammars that have sections devoted to partitives and, in a few cases, by searching for the relevant examples in the grammars (if the grammar lacked such a section or the section was not informative enough). Originally, many more grammars were consulted but the final selection ended up with 138 languages because many grammars were excluded as they did not provide sufficient information. In this sense it is a convenience sample. The 138 languages represent 46 families from all six macroareas.

There are somewhat more entries (171) than languages (138) because one language may have up to three different partitive constructions and thus entries. Lithuanian, for instance, attests two partitives: along with the possessive strategy (seen in (5) above), there are partitives based on the separative strategy with the preposition $i\check{s}$ 'from'. The two partitives are used somewhat differently. For example, the separative strategy of Lithuanian cannot be used as a pseudopartitive while the possessive strategy can. Another example of multiple entries for one language is the partitive marker de for the NP-internal syntactic type and the partitive pronoun en 'thereof' (the pronominal type) in French (see Section 4.2.1 below).



Figure 1: The 138 languages of the sample.

Not all examples that were translated with the English partitive (out) of were taken into account. For example, the two of us, both of them (often just rendering the respective dual forms), none of us or all of us were not taken into consideration. The motivation behind this decision was to exclude examples which seem to be partitives just due to the restrictions on numeral and quantifier modifiers in English translations.

Furthermore, in order to distinguish between implicit expressions such as *some flowers* in (11) and unmarked partitives (Section 4.2.2.5), only examples with the supersets explicitly marked as definite (pronouns, demonstratives, etc.) were considered as evidence of an unmarked partitive.

4 Constructional variation

In what follows, I present a systematic overview of different partitive constructions. I first discuss the variation in the degree of internal integration of partitive constructions (not included into the database) in Section 4.1 and then turn to the coding variation in Section 4.2.²¹

4.1 Degree of internal integration of the quantifier and the restrictor

From the semantic point of view, a true partitive requires at least two referents that may not be identical: the superset and the subset referent. Consequently, many languages map these two referents – iconically – into two interdependent constituents: one for the subset (*some* in (23)), patterning as the head, and one for the superset referent (*these students* in (23)), patterning as the dependent. This is the situation found in many European languages, for example, in English:

(23) some of these students

However, languages vary considerably in this respect. Even in English, the two constituents can be split (Hoeksema 1996a: 6), representing thus a somewhat looser syntactic integration:

²¹ Languages vary not only in terms of syntactic integration of the quantifier and the restrictor (Section 4.1), or in the variation of the marking strategy discussed in Section 4.2, but also in many other relevant respects such as the linear positions of the restrictor and the quantifier. For reasons of space, these cannot be discussed here.

(24) *Of the students, at least ten were drunk.*

Optional extraposition of the restrictor NP is also found elsewhere, e.g., in Ancient Greek, Yadu (Qiangic, Sino-Tibetan; LaPolla and Huang 2003: 101), Punjabi (Indo-European; Bhatia 1993: 189), Hinuq (Forker 2013: 723) or Lezgian (Nakh-Daghestanian; Haspelmath 1993: 97 ex. 195).

Moreover, in some languages, partitives may be expressed just by topicalizing the restrictor, e.g., in Jalkunan:²²

(25) Jalkunan (Mande)

[sìgì-sèè $m\'e?\bar{e}$ -nà- \grave{a}^n $b\'u?\acute{u}$ -n \acute{u}] $z\`ak\~u$ $f\~5^n$ [song-sing.NMLZ person-NOM-PL all-NOM] Z better '(Out of) all the singers, Zaki is better (=the best).' (Heath 2017: 262)

Partitives relying on topicalization of the restrictor NP represent rather loose structures with no strict syntactic dependency and integration, e.g., in Hinuq (Nakh-Daghestanian) (Forker 2013: 723). Examples in (26)–(29) illustrate this:

(26) Syer (Atlantic-Congo)

[mè pí plŏòb yìrè làŋ wò], yĕ plàà dà wồŋế
[but def children PRT in PRT] 2PLPRF INDEF put European
sèbé kẽde lẫŋ...
writing route in
'But, among the children themselves, have you let some embark on the

'But, among the children themselves, have you let some embark on the European school system (lit.: put into the route of European writing)...' (Dombrowsky-Hahn 2015: 299)

(27) Humburi (Songhay)

[háns-èy-nón gà:] í-fó: nèy nàmà [dog-DEF.PL-2SG in] ABS-one 1SG.OBJ bite 'One of your dogs bit me.' (lit. "in your dogs, one bit me") (Heath 2007: Section 5.4.10)

(28) Japanese (Japonic)

[Tyabudai-no asi]-wa ip-pon torete ita table-GEN leg-TOP one-CLF missing 'One of the portable table's legs was missing.' (adapted from Kim 1995: 226)

²² Only those superlatives have been taken into account in this study which were based on partitives morphologically.

In Jamsay, the subset NP even has to resume the kind of entities from the superset, which is a strong indication of the lack of any syntactic dependency relation between the two:

(29) Jamsay (Dogon)
gùgùn núŋô mà găn'n, gùgûn nù:yn éwé-ỳ
watermelon DEM POSS between watermelon five buy.IMPF-1PL
'We'll buy five of those watermelons.' (lit. 'Among those watermelons,
we'll buy five watermelons.')
(Heath 2008: 254)

By contrast, given that both the superset and the subset belong to the same kind and the subset referent is referentially anchored in the superset referent, some other languages have partitive expressions consisting of just one phrase. Thus, in Syer, the indefiniteness marker – also meaning 'some' – if added to a definite expression yields the true-partitive relation both with count and mass nouns:

(30) Syer (Atlantic-Congo)

mɛ̀ wò cɛ̃-plāa sɔ̀ nìwuru.

cons our woman-indef marry even

'... and even married some of our women.'

(Dombrowsky-Hahn 2015: 299)

Similarly, the indefinite pronoun *sekin* 'some' in Huasteca Nahuatl may just adjoin to the inherently definite personal pronoun in order to encode the true-partitive relation (the connector/relativizer *tlen* may be dropped)²³:

(31) Huasteca Nahuatl (Uto-Aztecan)
sekin (tlen) imowanti
some (REL) 2PL
'some of you'
(Beller and Beller 1979: 251)

While the true-partitive relation is expressed by a single phrase in (30) or (31), it can also be expressed just by a single word:

(32) Ute (Uto-Aztecan)

máamachi-u-a-tu(-mu)

woman-PL-POSS-NMLZ(-ANIMATE.PL)

'one (some) of the women'

(Givón 2011: 99)

²³ Note that the relativizer *tlen* need not be genuine in the partitive construction but due to the impact of Spanish (cf. Hill and Hill 1986: 292; Kurt Malcher, p.c.).

(33) Dhofari Arabic (Afroasiatic)

baΥð-hum

some-3PL

'some of them'

(Davey 2016: 102)

(34) Gitxsan (Tsimshian) (Bicevskis et al. 2017: 323)

hlagats'uu-diit
some-3PL
'some of them', 'others of them'
(Bicevskis et al. 2017: 323)

There is thus quite a lot of variation in the degree of integration of the restrictor with the quantifier from rather loose syntactic relation under extraposition to highly integrated, one-word expressions.

4.2 Coding variation of partitives

In this section, I explore the coding patterns to mark partitive constructions as well as their areal biases along two variables: syntactic *type* (Section 4.2.1) and marking *strategies* (Section 4.2.2). Table 3 presents the overview.

4.2.1 Syntactic types

The first variable is the syntactic *type*, i.e., whether partitives are encoded *NP-internally* or *pronominally*. Note that, if the partitive construction is morphologically marked by zero (Section 4.2.2.5), one can also speak about the third syntactic type of partitive constructions, i.e., *juxtaposition*.

NP-internal markers are primarily flags (adpositions and case inflection) but also, rarely, possessive indexes. For example, in (35), the ablative case marker -*duk* is an NP-internal partitive marker:

Table 3: Coding variation of partitives.

Туре		Strategy			
NP-internal pronominal	possessive	separative	locative	unmarked	other
	possessive	separative	locative	unmarked	other

(35) Evenki (Tungusic)

Bi ēkun:ma-da tari-l-duk d'eptile-1-duk

1SG anything-ACC that-PL-ABL(=PART) food-PL-ABL(=PART)

e-che-v gara.

NEG-PST-1SG take

'I took nothing of that food.'

(Nedyalkov 1994: 26)

The second syntactic type, i.e. the *pronominal* type, comprises two different subtypes, which, however, are grouped together here for practical reasons. Their common property is their position: in contrast to the NP-internal partitive markers, partitive pronominals occur predominantly outside NPs. In most of the cases, these are highly integrated into the verbal complex, sometimes even phonetically, i.e., via cliticization. This is the case in many Bantu languages in which partitives are encoded by means of class markers (classes 16–18) that otherwise index locations:

- (36) Oshikwanyama (Atlantic-Congo, Bantu)

 Opún' óu-leke, ito-kúfá=mo?

 be 14-candy NEG.PRS.2SG-take=LOC(=PART)

 'There is candy, why don't you take some?'

 (Halme-Berneking 2017: 147)
- (37) Luganda (Atlantic-Congo, Bantu)

 Aguzee-ko bi-satu

 3SG.buy.PST-PART CL-three

 'He has bought three of them.'

 (Ashton et al. 1954: 420)

The pronominal type combines the following subtypes: *partitive pronouns, partitive quantifiers* and *partitive particles*. The first two are lexically marked partitives. However, there is an important distinction between the two: only partitive pronouns pronominalize the superset and thus have the meaning 'of them/of it', cf. German *davon* 'of it/them'. By contrast, partitive quantifiers have the meaning 'some of', 'a few of' and the superset has to be expressed by an NP. Partitive particles, in turn, behave very much like partitive quantifiers, although they do not originate from quantifiers. I discuss partitive pronouns and quantifiers in more detail.

Partitive pronouns often carry no partitive marking that is found outside the pronominal paradigm in the language, and may be considered as being lexically marked as partitives. For example, there is the partitive pronoun en in French

(cf. Ihsane 2013), *ne* in Italian, *er* in Dutch and different pronouns in German dialects (cf. Strobel and Glaser 2019) – all can pronominalize only third-person supersets:

(38) French

J 'en ai vu deux

1SG PART AUX.1SG see.NMLZ two

'I have seen two of them.'

(adapted from Ihsane 2013: 241)

Partitive pronouns are also found outside of Europe, in less familiar languages. For example, there is a clitic third-person partitive pronoun -i'ij in Itzaj or aa in Tlingit:

(39) Itzaj (Mayan)

Yan in-jan-t-ik-i'ij?

OBLIG 1SG.A-eat-TRN-IIS-PART

'Do I have to eat some of this?'

(Hofling 2000: 251)

In contrast to French *en* or Dutch *er*, which are inherently third person only, some languages have partitive pronouns that may occur in all three persons. In Eibela, there is even a whole set of partitive pronouns inflected for person: 1st *ni:jɛ:*, 2nd *gi:jɛ:*, 3rd animate *i:jɛ:* (Aiton 2016: 117). These are evidently derived from the plural forms of the respective persons by adding the marker -:*jɛ:* (homonymous with a locative marker). Thus, the first person *ni-:jɛ:* can be analyzed as 1PL-LOC(=PART). These pronouns may only be used in the subject or in the object slot in this language:

- (40) Eibela (Bosavi; Papua New-Guinea)

 ge:sa je:niba kowa:bo i:je: e:na: de: gi:je: hene sene=jo:bo:

 name name name 3.PART DEM fire build.fire DUR stay.PST=INF

 'Gisa, Jennifer, and Kowa:bo, some of them had made a fire and were

 sitting there.'

 (Aiton 2016: 119)
- (41) Eibela (Bosavi; Papua New-Guinea)

 nɛ:na: i:jɛ: o:-mɛ:na:

 1DU 3.PART shoot-FUT.1

 'We two will shoot at them.'

 or 'We two will shoot some of them.'

 (Aiton 2016: 119)

Similarly, the partitive-paucal marker $-s \grave{e} \eta$ attaches to personal pronouns in Wadu Pumi, yielding thus partitive pronouns inflected for all three persons: $t \grave{e} -b \grave{u} - s \grave{e} \eta$

(3-COLL-PART) 'one/some of them', $ni\eta$ - $b\acute{u}$ - $s\grave{e}\eta$ (2-COLL-PART) 'several of you', $i\eta$ - $b\grave{u}$ - $s\check{e}\eta$ (1-COLL-PART) '(five) of us' (Daudey 2014: 86, 151, 579, 585):

(42) Wadu Pumi (Sino-Tibetan; Qiangic) \dot{e} - $b\dot{u}$ - $s\dot{e}\eta$ $k^h\dot{\partial}$ - $\varphi\dot{\partial}$ 1-COLL-PART OUT-go '(...) several of our household went (...)' (Daudey 2014: 205)

While in Eibela or Wadu Pumi, partitive pronouns are syntactically free in that they need not occur next to the predicate, other languages such as French show the tendency for the partitive pronouns to occur immediately next to the verb and often to cliticize with it.

Partitive quantifiers ('some of'), by contrast, typically do not pronominalize the superset (the kind with pseudo-partitives); the latter is expressed by an NP, cf. exes. (43)–(46):

- (43)Apma (Austronesian; Vanuatu) Ва naanong, sawiri=te bwet si=ah. ba na=ma COMM now COMM 1SG=PRSP grate=PART taro POL=DEM 'But now, I'll grate some of this taro first.' (Schneider 2010: 166)
- (44) Caquinte (Arawak; Peru)

 a-N-pišoNka-yaraNt-ak-e-ro aiNtočapaki

 1INC-FUT-put-PART-PERF-FUT-3F type.of.yucca

 'We will put some of the yucca.'

 (Townsend et al. 1988: 77)
- (45)Tlingit (Athabaskan-Eyak-Tlingit; USA) <u>k</u>wáa<u>k</u>t tsé aa néi x'wán yeehwáan amiss PRT PART happen IMPV 2PL 'don't let something happen to any of you' (Crippen 2012: 327)
- (46) Cherokee (Iroquoian; USA)

 tsò? ij-àni:-2-â: ts-ùw-ĕ:tsi:

 three PART-3PL-be.that.many.PERF PL-3SG-child

 'three of his sons'

 (adapted from Julian 2010: 48)

Partitive pronouns and partitive quantifiers have often quite different historical sources: while partitive pronouns often develop from deictics like 'there' or 'from there/it', partitive quantifiers stem from existential quantifiers such as 'some' or 'a few'.

Even though the distinction between partitive pronouns and partitive quantifiers seems to be so far straightforward, in practice, it is often difficult to discern whether the pronominal is inherently a partitive pronoun or a partitive quantifier or a partitive particle that occurs in a clause with object drop as in (47):

(47) Hdi (Afroasiatic, Chadic; Africa)

ndà-'á-ndà

swallow-PART-swallow

'he swallowed some of (them)'

(adapted from Frajzyngier 2002: 264)

Note that the pronominal type is orthogonal to the marking strategies to be discussed immediately below (Section 4.2.2) since partitive quantifiers and pronouns may also reflect different marking strategies such as the separative strategy with the German davon 'from it/them' (from the pronominal da(r) and the adposition von 'from') or the possessive strategy with the Dutch er 'of it/them' (historically from a genitive pronoun) or the unmarked strategy (e.g., tuut 'some' in Avava). Having said that, most of the time, it is very difficult to determine the marking strategy of the partitives of this type.

Finally, there is the third subtype *partitive particles* which is even more difficult to distinguish from partitive quantifiers. Partitive particles are found, for example, in many Bantu languages, cf. the class 18 index *=mo* in Oshikwanyma:

(48) Oshikwanyma (Bantu)

Opú-n' óu-leke, ito-kúfá=mo?

is 14-candy NEG.PRS.2SG-take=18(=PART)

'There is candy, why don't you take some of (it).'

(adapted from Halme-Berneking 2017: 147)

Crucially for the pronominal type that all its subtypes – *partitive pronouns*, *partitive quantifiers* and *partitive particles* – are used pronominally in that either the quantifier or the restrictor or both are not expressed as arguments.

4.2.2 Marking strategies

The second variable, *marking strategy*, tags the underlying metaphor for the morphological marking of partitives: the possessive strategy (Section 4.2.2.1), the separative strategy (Section 4.2.2.2), the locative strategy (Section 4.2.2.3), some other, infrequent strategies (Section 4.2.2.4), and finally the unmarked strategy

(Section 4.2.2.5) that relies on just juxtaposing the quantifier and the (definite) restrictor NPs.

4.2.2.1 Possessive strategy

Languages that employ the possessive strategy exploit the internal-possession metaphor for marking partitives (Hoeksema 1996a: 6; Koptjevskaja-Tamm 2001, 2009; Luraghi and Kittilä 2014: 55). Lavukaleve is such a language (Terrill 2003: 94):

(49) Lavukaleve (isolate)

Ma-fan e-su-ne

3PL.POSS(=PART)-some 3SG.N.OBJ-tie-IMPF

'Some of them were tying it up.'

(Terrill 2003: 64)

In (49), the restrictor is coded by the possessive plural pronominal prefix *ma*-while the quantifier is *fan* 'some'. While Lavukaleve employs possessive indexes, a number of Indo-European languages employ flags (the genitive case) to encode partitives under the possessive strategy, e.g., Lithuanian, Russian, Punjabi or Armenian.²⁴ However, the possessive strategy is recessive in many Indo-European languages and tends to be replaced by the separative strategy to various degrees.

4.2.2.2 Separative strategy

Another very frequent strategy to encode the partitive relation is to use a spatial periphrasis based on the metaphor *from*-Ground (Hoeksema 1996a: 6; Koptjevskaja-Tamm 2001, 2009; Luraghi and Kittilä 2014: 55). For example, nearly all Turkic languages employ the ablative case to encode partitives, including pseudopartitives (cf. Uzbek in Bodrogligeti 2003: 111). Similarly, nearly all Semitic languages may employ the preposition *min* (and its etymological variants) for marking partitives. Below I list some examples of the separative strategy: (50) with an explicit quantifier and generalized partitives in (51)–(53):

(50) Tuvaluan (Austronesian)

E toko lua i tamataene ne mmai o fesoasoani NONPST NUM two **from** young-man NONPST come and help

mai ki au.

ADV to 1SG

'Two of the young men came and helped me.'

(Besnier 2000: 330)

²⁴ I tagged Armenian as also having the possessive strategy although the relevant function is encoded by the dative. The reason for this is that the Armenian dative can also express possession.

- (51) Tyvan (Turkic) *šay-dan ižer men*tea-ABL(=PART) drink.P/F 1SG

 'I'll drink some (of the) tea.'

 (Anderson and Harrison 1999: 16)
- (52) Imonda (Border)

 ka-ia-nèi-m ainam iaha-t

 1-LOC-ABL(=PART)-GL fast die-CF

 'one of us would die quickly'

 (Seiler 1985: 58)
- (53) Hungarian (Uralic)

 Ettem a kenyer-ből

 eat.PST.1SG DEF bread-ABL(=PART)

 'I was eating / ate some of the bread'

 (Moravcsik in Koptjevskaja-Tamm and Wälchli 2001: 664)

This strategy is also found in Africa. For example, the ablative case is used with pronouns and with nouns headed by numerals in Wolaytta (Na-Te-Omotic; Lamberti and Sottile 1997: 216). Some Bantu languages employ the locational class markers otherwise used to encode the source. Compare the class marker =mo (class 18) that is used for the ablative meaning in its spatial use in Kirundi (Devos et al. 2017: 73) or Oshikwanyma (Halme-Berneking 2017: 140):

(54) Oshikwanyma (Bantu)

Opú-n' óu-leke, ito-kúfá=mo?

is 14-candy NEG.PRS.2SG-take=18(=PART)

'There is candy, why don't you take some of (it).'

(adapted from Halme-Berneking 2017: 147)

According to Halme-Berneking (2017: 147), the class marker 18 -*mo* signals source and thus partitive interpretation 'some of (it)'.

Note that the separative strategy can be further subdivided into subtypes such as 'from-in', 'from-at' or 'from-among', cf. the variation of partitive markers in Romanian: *din* 'from' (consisting of *de* 'of, from' and *in* 'in') vs. *dintre* 'from, among, between' (*de* 'of, from' and *între* 'among') (Tănase-Dogaru 2017). Unfortunately, my data is not fine-grained enough to consistently distinguish between these subtypes (the same holds for the locative strategy, see Section 4.2.2.3 below).

Sometimes the distinction between the separative and the locative strategy (Section 4.2.2.3) is obscured by case stacking. In Punjabi, the partitive marker combines both the locative and the ablative strategy in that it employs the post-position $vicc\bar{o}$ 'among', which is morphologically complex containing the post-position vicc 'in' and the ablative marker $t\bar{o}$ 'from':

(55) Punjabi (Indo-European, Indo-Aryan; India)

ó de muNDiãã viccō tinn amriikaa gae.

3SG.M GEN.M.PL boys.OBL among(=PART) three America go.PST.M.PL

'Among his boys, three went to America.'

(Bhatia 1993: 189)

Here, it is reasonable to classify the partitive according to its outermost marker, i.e., the separative $t\bar{o}$ 'from'. Similarly, the partitive case in some Nakh-Daghestanian languages, for example, in Archi, Khwarshi (Kibrik 1977: 174; Khalilova 2009) or in Hinuq (-t-es '-in-from', Forker 2013: 83) stems from an earlier inter-elative.

Finally, it is not entirely clear what the spatial origin of the partitive case (in -DA) in the northern Siberian Turkic languages such as Yakut or Tofa is. While it goes back to the Proto-Turkic locative case (Nevskaya 2017: 278; Ubrjatova et al. 1982: 134), separative uses of this case are also attested in Old Turkic, cf. *tengri-de* /sky-da/ 'in the sky, from the sky' (Erdal 2004; Nevskaya 2017: 279; Ubrjatova et al. 1982: 134).

Within the NP-internal syntactic type, it is the separative strategy that most often develops into a dedicated partitive marker (partitive case) while losing its original spatial meaning. A dedicated partitive case stemming from an ablative is found, for example, in all Finnic languages. In Kryz (Nakh-Daghestanian), the meaning of the subelative case came to be exclusively used for partitive functions while its spatial meaning has been taken over by a new postposition (Authier 2009: 82). In Archi and in Khwarshi (Nakh-Daghestanian), the dedicated partitive case stems from an earlier inter-elative (Khalilova 2009; Kibrik 1977: 174). Note that the pseudo-partitive use of the dedicated partitive case is prevalent in all these languages.

4.2.2.3 Locative strategy

The locative strategy implies that the partitive marking is syncretic with a location marker. Somewhat unfamiliar in this context is the English conative construction with *at* (Levin 1993: 6) sometimes with *on* (Levin 1993: 43):

- (56) a. Margaret cut the bread
 - b. Margaret cut at the bread

The insertion of the preposition at in (56) yields the true-partitive interpretation of the object, similar to *Margaret cut some of the bread*.

The locative strategy is marginally also found in German with the prepositions *an* 'at' (cf. Filip 1999: 267) and *unter* 'among'.

(57) German (Indo-European; www²⁵)

Lass	Deinen	Mann	doch	mal	am	Fisch probieren
und	beurteilen,	ob's	salzig	ist	at.DEF.DAT.SG	-
and	judge	whether it	salty i	S		

^{&#}x27;Let your husband taste the fish and tell whether it is salty.'

Otherwise German primarily employs the separative strategy with *von* 'from' or the possessive strategy with the genitive.

Outside of Europe, the locative strategy is frequently found in Africa, especially in various Bantu languages (cf. Persohn and Devos 2017). Many Bantu languages such as Soga, Kwanyama, Ruund, Lega, Haavu, Luvale, Ganda, Nyakyusa or Umbundu employ place class markers (class 16 and 17) which often occur as verbal clitics (subsumed under the pronominal type here) to encode partitive functions (Persohn 2017a; Persohn and Devos 2017: 21–24). For example, the location marker *îbè/îbé* is used in Oko (Atoyebi 2010: 131):

(58)Oko (Atlantic-Congo; Nigeria; Atoyebi 2010: 132) égbén ábe íbè уò úbó òóre one children DEF.PL LOC go house 'One of the children went home.' (Lit. One inside/among the children went home.)

In Luvale, the place class markers *ku* (class 17) is used on the object NP 'food' (which is class 15 otherwise):

²⁵ http://www.gesundehunde.com/forum/archive/index.php/t-87252.html.

(59) Luvale (Bantu)

eji ku-ly-anga ku-ku-lya c-ami

1²⁶.AUX INF-eat-HAB 17-15-food 15-POSS.1SG

'He eats of my food.'

(Persohn and Devos 2017: 22)

Numerals and quantifiers in a number of Dravidian languages such as Malayalam or Tamil are used as modifiers of the NP (cf. English *two boys*). They may also be used pronominally – with a classifier-like noun, cf. *peer* 'person' in (60) – resulting in the true-partitive meaning. In this case, they follow the dependent NP and require the locative case on it (Asher 1985: 116; Asher and Kumari 1997: 218–219):

(60) Malayalam (Dravidian; India)

aankuttikalil rantə peer vannu

male.child.PL.LOC two person come.PST

'Two of the boys came.'

(Asher and Kumari 1997: 218)

The locative strategy is different from the separative or possessive strategies in that it might require that the subset be located at the same place as the superset, at least, at an early stage of grammaticalization into a partitive marker. Thus, this strategy is less suitable with predicates that denote separation, e.g., *to leave*. Contrast (61) or (63) vs. (62):

- (61) Some among us have achieved great things.
- (62) Some among us have left our town. (?)
- (63) Some of us have left our town.

Generally, the locative strategy subsumes two subtypes. The first subtype is based on the concept of 'among' (e.g., in Togo Kan, cf. Heath 2015: 150, Section 8.2.12):

(64)Togo Kan (Dogon); í'n nàŋá kè kénè, nàŋá à:ŋá nù-y cow how.many cow 1SG.POSS POSS among(=PART) die-PFV 'Among my cows, how many cows died?' (Heath 2015: 289, Ex. 418–419, Section 13.2.7)

The second subtype is based on the spatial concept of containment 'inside' (Luraghi and Kittilä 2014: 54–55), for example in Koyra Chiini (Heath 1999: 105), or proximity, as, marginally in English (56) or German (57), for example. Some

²⁶ The first class among the third persons – usually singular humans.

languages employ both subtypes at the same time, for example, Jamsay (Dogon) marks the superset NP with either $b\grave{e}r\hat{e}$: 'in' or with $g\check{a}n\grave{n}$ 'between' (Heath 2008: 471). In other languages, these subtypes are difficult to distinguish. For example, in Biblical Hebrew the locative/instrumental preposition b- occurs with human referents in the plural in the meaning 'among' (65); Homeric Greek is similar (66):²⁷

- (65) Biblical Hebrew (Ct 1.8.; Semitic)

 hay-yāfā b-annāšīm

 DET-beautiful in-women

 'beautiful among women'

 (Pennacchietti 1974: 175)
- (66) Ancient Greek; Indo-European)

 ariprepéa Trốessin

 beautiful.NOM.SG.F Troans.DAT=LOC.PL

 'the beautiful [one] among the Troans'

 (Homer Iliad 6.477)

Furthermore, partitive pronouns (Section 4.2.1 above) sometimes stem etymologically from locative pronouns such as 'there'. For example, -iij in Itzá (Mayan) is homophonous with the locative pronoun -iij 'there' (Hofling 2000: 304, 306) and therefore it probably descends from it historically. Similarly, the partitive pronouns in Eibela (Bosavi) are all derived by means of the marker $-j\epsilon$:, which is homonymous with the locative marker and seems to stem from it.

Some languages have general spatial adpositions which encode different kinds of spatial relations including the movement from the Ground, rest inside the Ground, etc. In Mussau (Austronesian; not included in the database), the partitive is coded by the proclitic eta= which has a very general locational meaning including both the separative and the locative meanings and which, moreover, requires the possessive marker as well (cf. Brownie and Brownie 2007: 152–3). Similarly, the preposition $m\dot{\psi}$ in Degema (Atlantic-Congo) can occur as locative, ablative, etc. (Kari 2004: 213; tagged as unclear in the database).

4.2.2.4 Rare strategies

I have only one example of a language in my database that employs a serial verb: Neverver has the partitive serial verb -dor (Barbour 2012: 221):

(67) Neverver (Austronesian; Vanuatu)

Na ni-khan-dor ni-kkan-ian na.

²⁷ The possessive strategy is more typical for Ancient Greek.

```
1SG 1SG.REAL-eat-PART NMLZ-eat-NMLZ 1SG 'I partly ate my food.'
(Barbour 2012: 221)
```

According to Barbour (2012: 325), this partitive marker stems from a verb *dor* 'become thin, lean'. However, it is not excluded that this marker was in fact originally an incorporated quantifier/pronoun, just synchronically patterning similar to a serial verb.

Another rare example is Apatani (Sino-Tibetan), which marks the subset quantifier by means of the collective marker xi on the numeral or quantifier (Abraham 1985: 59):

- (68) Apatani (Sino-Tibetan)

 hime hine xi

 boys three COLL(=PART)

 'three of the boys'

 (Abraham 1985: 59)
- (69) Apatani (Sino-Tibetan; India))

 mólu tane xi

 3PL some COLL(=PART)

 'some of them'

 (Abraham 1985: 59)

Another rare strategy is based on the comitative/instrumental marking of the superset NP. It is marginally found in Scandinavian languages (cf. Koptjevskaja-Tamm 2001). Further examples include Jalkunan (Mande), cf. (25) above, Mapdungun, or Ulwa (which employs the comitative enclitic =ul):

- (70) Ulwa (Ulmapo; Papua New Guinea)

 nï lïmndï tïn ndï=ul kuma ndï=ala

 1SG eye dog 3PL=with(=PART) some 3PL=for

 'I saw some of the dogs.' (lit. 'I saw some with the dogs.')

 (Barlow 2018: 200)
- (71) Mapudungun (Araucanian; Chile)

 itro-kom-mew wente-le-lu
 quite-all-INS(=PART) top-ST-SVN

 'the topmost of all'
 (Smeets 2008: 218)

CORRECTED PROOF

DE GRUYTER MOUTON

Typology of partitives — 33

(72) Mapudungun (Araucanian; Chile)

iñchiu misha-w-a-y-u korü-mew
 1DU share.with-REF-NRLD-IND.1NS soup-INS(=PART)
 'we will eat some of the soup together'
 (Smeets 2008: 218)

Finally, some languages regularly employ lexical items to encode the partitive relation. For example, Madurese (Austronesian) has the noun *sabagiyan* 'part' (sometimes with the separative preposition *dhari* 'from') to convey the meaning of an indefinite subset (Davies 2010: 198):

(73) Madurese (Bosavi)

Sabagiyan mored-da Pa' Hadi bakal lulus tes. part(=PART) student-DEF Mr Hadi will pass test 'Some of Pak Hadi's students will pass the test.' (Davies 2010: 198)

4.2.2.5 Unmarked strategy

Quite a number of languages allow the true-partitive relation to be morphologically unmarked and thus employ the strategy that is otherwise used for the plain – i.e., non-proportional – quantification such as 'many books' or 'five letters', cf. Tuvaluan:

(74) Tuvaluan (Austronesian: Tuvalu)

E toko lua tamataene ne mmai o fesoasoani mai ki au. NONPST NUM two young-man NPST come and help ADV to 1SG 'Two young men came and helped me.'

or: 'Two of the young men came and helped me.'

(Besnier 2000: 330)

The unmarked strategy is different from implicit expressions (Section 2.2.2) such as *some flowers* in (11). While, with implicit expressions, the meaning of the true-partitive relation is produced by the discourse situation and not by encoding it because the superset has to be retrieved from previous discourse, the unmarked strategy is found when both the subset and the superset are explicitly mentioned. Since many languages do not have dedicated definite determiners that would disambiguate a definite NP, in practice, an unmarked strategy can only be safely assumed if an indefinite expression (an indefinite-plural marker, indefinite pronoun or quantifier) is found to modify an NP that is explicitly marked as definite

34 — Seržant

DE GRUYTER MOUTON

(e.g., by a determiner, demonstrative, definite pronoun, etc.). Compare the following examples:

- (75) Mapudungun (Araucanian; Chile)

 kiñe-ke ñi pu wenüy

 some-DISTR POSS.1SG PL friend

 'some of my friends'

 (Smeets 2008: 136)
- (76)Avava (Austronesian; Vanuatu) mwiniel Komat-yan i moroko-n tuut ier rib-3sg 1PL.EXC.R-eat taro INS some PL 'We ate the taro with some of its ribs.' (Budd 2014: 553)
- (77) Avava (Austronesian; Vanuatu) *I-yan tuut emer ki*3SG.R-eat some eel DEM
 'He ate some of the eel.'

 (Budd 2014: 553)
- (78) Dholuo (Nilotic; Sudan)

 Nyithinda moko odhi sikul

 children.1sg.poss some go-pfv school

 'Some of my children have gone to school.'

 (Okoth Okombo 1997: 24)
- (79) Supyire (Atlantic-Congo; Mali)

 wùu wà

 1PL INDEF.SG

 'one of us'

 (Carlson 1994: 195)
- (80) Movima (isolate; Bolivia)

 us al-kaka:-ye

 DEF.M similar-many-person

 [lit.] 'the male (one out of my) relatives'

 (adapted from Haude 2006: 436)

Some languages distinguish between definite and indefinite plural markers. The latter can sometimes be used to encode the true-partitive relation:

(81) Syer (Atlantic-Congo)
 mè wò ċĕ-plàà sò nìwurủ.
 CONS our woman-INDEF marry even
 '... and even married some of our women.'
 (Dombrowsky-Hahn 2015: 299)

Since the semantic dependency between the superset and the subset is encoded by the very general strategy of juxtaposition, this strategy does not differentiate morphologically between the true-partitive relation and a wide range of other functions that the bare juxtaposition of two phrases may encode in a particular language. ²⁸ Therefore, languages with the unmarked strategy often have another strategy that is based on a particular marker. For example, Imbabura Quechua or Tuvaluan may also express the true-partitive relation with a separative adposition (cf. *i, mai* or *mai i* in Tuvaluan) in addition to the unmarked strategy in order to exclude any non-true-partitive interpretations.

4.3 Synopsis of different syntactic types and strategies

Below I present the quantitative evidence on different syntactic types (Section 4.2.1) and marking strategies (Section 4.2.2).

4.3.1 Synopsis of different syntactic types

Table 4 provides the frequencies of different syntactic types: pronominal, NP-internal and no marking at all.²⁹ In order to level out the impact of larger families, the totals of families per type are provided first. Here, one family can only contribute one point per type regardless how many languages of that family attest the type (if a family attests several types, it contributes a point to each of those):

²⁸ There might be some other, non-morphological means to differentiate between the two (such as the order of elements or words) but these cannot be discussed here and would require a typology of their own.

²⁹ Since the unmarked strategy (Section 4.2.2.5) only relies on juxtaposing the subset and the (definite) superset NPs, it can theoretically be analyzed as NP-internal marking. However, I refrain from doing so and tag all partitives of this strategy with the non-applicability tag in the syntactic type.

Table 4: Frequencies of the different syntactic types.

	NP internal (flag or possessive index)	Pronominal (lexically marked)	Unmarked
Per family	28	11	26
Per language	104	25	43

While the pronominal type is much less frequent, the NP-internal marking is certainly the preferred syntactic type:

(82) Syntactic type preference *NP-internal type > pronominal type*

This preference might even be stronger if one counts the juxtaposition – i.e., the unmarked – type as an NP-internal syntactic type.

4.3.2 Synopsis of the different marking strategies

Before I turn to the quantitative evidence, note that languages sometimes allow for more than one strategy (e.g., the separative *von* 'from', the locative *an* 'at' and *unter* 'among' in German), often with different frequencies and degrees of grammaticalization – something that cannot be captured within the frame of this study.

Furthermore, since my aim is to establish the underlying metaphor and thus the source of partitive markers, the classification is unavoidably somewhat diachronic. Thus, partitive markers that synchronically have different meaning facets in the language are tagged according to the function that the partitive is most probably derived from. For example, the partitive marker *de* in French functionally matches two of the strategies listed above: the possessive and the separative strategy. However, since the partitive function historically derives from the separative and not from the possessive function as is evidenced by Latin (Carlier and Lamiroy 2014), French is tagged as having the separative strategy (the same holds for the partitive pronoun *en* 'thereof'). If the grammar does not provide any diachronic information on the ambiguous marker, the partitive is tagged according to the least abstract meaning of the marker in the given language along the following scale: separative < locative < possessive. This seemed to be a reasonable convention that would only affect very few entries anyway. In some instances, the strategy was tagged as *unclear*.

Table 5 provides the absolute frequencies of each strategy in the database. These counts show that none of the marked strategies is universally preferred because different strategies are found to occur with comparable frequencies.

CORRECTED PROOF

DE GRUYTER MOUTON

Typology of partitives — 37

Table 5: Frequencies of the different marking strategies.

	Possessive	Separative	Locative	Unmarked	Unclear ³⁰	Other
Per family	11	20	13	26	10	5
Per language	24	56	25	43	19	5

Having said this, there are some clear preferences. While, e.g., the instrumental strategy is clearly dispreferred (subsumed under *other* in Table 5), the separative and the unmarked strategy are more frequent than any other strategy. I summarize:

(83) Strategy preferences (the preferred strategy is to the left) unmarked > separative > locative/possessive > instrumental

The difference between the locative/possessive and the separative strategy is most probably due to areal pressures that boost up the absolute numbers of my sample, which is biased for Eurasia (Section 5).

5 Areal distribution

The marking strategies discussed above reveal areal biases. Figure 2 suggests that the separative strategy is typical for Eurasia, while the locative strategy is more typical for Africa.³¹ Table 6 presents the row counts and highlights the outliers.

Although the sample is not large enough to provide for a more precise picture, it is sufficient to make two claims. First, the distribution clearly shows that none of the marked strategies is universally preferred. Secondly, the null hypothesis of the marking strategies being normally distributed across all macro-areas can be rejected. Taken together, this means that strategy selection is subject to areal pressures.

³⁰ Pronominal partitives are mostly morphologically intransparent, identifying the marking strategy is therefore often impossible. For this reason, pronominals are often tagged as *unclear* in the database.

³¹ This, of course, does not exclude that these strategies may be found elsewhere. For example, the locative strategy may also be found in Samoan (Austronesian; not included into the database), cf. the locative preposition i that may be used as a generalized partitive in this language (cf. examples in Mosel and Hovdhaugen 1992: 108).



Figure 2: The geograpic distribution of the marking strategies³².

Table 6: Coding strategies across the macro-areas, per languages, bold indicates statistical significance.³³

	Africa	Australia	Eurasia	North America	Oceania	South America
Separative	3	2	44 (57%)	_	3	3
Possessive	1	1	17 (22%)	1	4	-
Locative	15 (56%)	-	8 (10%)	_	1	1
Unmarked	8 (30%)	8	8 (10%)	5	10 (50%)	4
Other	_	-	1	_	2	2
	27 (100%)	11	77 (100%)	6	20	10

To test this I computed the standardized residuals in the *chi*-square test.³³ The separative strategy in Eurasia and the locative strategies in Africa have revealed themselves as statistically significant outliers (standardized residuals < 2.5). Given that the underlying sample is a convenience sample, I have also computed Fisher's

³² Note that more than one partitive strategy in one language may not be properly represented on the map for technical reasons.

³³ Statistically significant means the standardized residuals for the outlier cells in bold were above/below 2.5 in a chi-square test. Despite the fact that the underlying dataset is a convenience sample, the test is a legitimate method because the availability of a good grammatical description does not interfere with particular strategies of partitive coding; the sample of partitive coding strategies can thus be considered to be picked out randomly and thus allow for statistical inferences on the entire "population".

Table 7: Contrasting the separative strategy in Eurasia with els

	Eurasia	Elsewhere
Separative	44	11
Other	33	63

Exact test in a two-by-two table (Table 7, p < 0.001) that puts fewer requirements on the distribution of the observed data (Janssen et al. 2006: 425).

The areal bias towards the separative strategy in Eurasia is not only supported by genealogically unrelated languages of this area selecting this pattern (e.g., Turkic, Basque, etc.) but also by those languages of the sample that are related. These languages too provide genealogically independent evidence for the areal bias. For example, many modern European languages have partitive adpositions that stem from ablative adpositions such as French *de*, Dutch *van* or Latvian *no*. Crucially, despite the fact that these languages all belong to the same stock (Indo-European), this corelation is not due to common inheritance, because these adpositions are demonstrably not interrelated etymologically but represent innovations across and even within the subfamilies. By contrast, the original, inherited pattern of Indo-European – the possessive strategy (by means of the genitive case) – is recessive in all branches of this family, albeit to a different degree.

For example, Latvian (Baltic, Indo-European) employs the preposition *no* 'from' while the closely related Lithuanian uses the preposition *iš* 'from' and not the etymologically related preposition *nuo* 'from'. This means that the two Baltic languages innovated independently but under the same areal pressure. Similarly, although all Germanic languages rely on the separative pattern, the morphological markers are not etymologically interrelated: cf. Scandinavican *av/af* 'from' vs. German *von* 'from'. In the same vein, some Slavic languages employ unrelated separative prepositions to encode partitives, cf. Serbian *od* 'from' vs. Russian *iz* 'from'.

The fact that the correlation is only structural but not etymological, even among closely related languages, suggests that the bias for the separative strategy is motivated by areal pressure, i.e., ultimately by language contact, and not by a commonly inherited pattern, even within Indo-European. Pattern borrowing of the separative strategy has also been documented for other families of Eurasia. For example, it has been explicitly argued for Evenki (Tungusic), Dolgan and Sakha (Turkic) in Pakendorf (2010: 727–729).

The separative strategy outside of Eurasia is also in some cases the result of language contact with some European language. For example, a number of

Mesoamerican and South-American languages copy the Spanish preposition *de* to mark partitives in terms of pattern and sometimes also in terms of matter borrowing (e.g. in Nahuatl, cf. Hober 2020).

More generally, there is a bias towards NP-internal marking, i.e. most typically by adpositional markers (separative, locative and possessive), and against the unmarked strategy in Eurasia, see figures in Table 8.³⁴ The latter finding corroborates the evidence from European languages in Koptjevskaja-Tamm (2009).

Table 8:	Places of marking	across the macro-areas	, bold are	statistically significant.
----------	-------------------	------------------------	------------	----------------------------

	Africa	Australia	Eurasia	N. America	Oceania	S. America
NP-internal	16	3	69	1	9	5
Pronominal	6	1	5	3	7	3
Unmarked	8	8	8	5	10	4

6 Other related functions

In Section 4 and Section 5, I have explored the coding patterns of partitives as well as their areal biases. This section is devoted to exploring functions that partitives may develop in addition to the expression of the true-partitive relation. This quest is inspired by the partitive case in Finnic languages which has prominently developed functions in the following domains: plain quantification (pseudo-partitives, Section 6.1, cf. (84b), hypothetical events (Section 6.2, cf. (87b), predicate negation (Section 6.3, cf. (85)), event quantification and aspect, cf. (86a) vs. (86b), (repeated from (1), (2), (3) and (4) above) and even politeness (Section 6.4):

(84) Finnish (Uralic)

Sö-i-n puuro-a eat-PST-1SG porridge-PART.SG

- a. 'I ate some of the porridge.'
- b. 'I ate some porridge.' (Huumo 2010: 91)

³⁴ In contrast to the distributions in the other areas, Eurasia has more than 15 times more NP-internal marking than marking via a partitive pronominal and ca. 10 times less the unmarked strategy than NP-internal marking. The chi-square test shows statistical significance (standardized residuals more than +/-2.5) for the interaction of Eurasia and NP-internal encoding (though this result might need additional verification given the low values of some other cells).

CORRECTED PROOF

DE GRUYTER MOUTON

Typology of partitives — 41

(85) Finnish (Uralic)

En voidel-lut suks-i-a.

NEG.1SG wax-PST.CONNEG ski-PL-PART

'I did not wax (the) skis.'

(Huumo 2010: 95)

(86) a. Lyhens-i-n hame-tta

shorten-PST-1SG skirt-PART.SG

'I was shortening the skirt.' or 'I shortened the skirt [somewhat].'

b. *Lyhens-i-n* hamee-n shorten-PST-1SG skirt-ACC.SG³⁵ 'I made the skirt short.'

(87) Finnish (Uralic)

a. Luul-i-n häne-t viisaaksi. think-PST-1SG 3SG-ACC wise 'I thought him to be wise.' (Huumo, p.c.)

b. Luul-i-n hän-tä viisaaksi.
think-PST-1SG 3SG-PART wise
'I (mistakenly) thought him to be wise.' (but he turned out not to be)

In this section, I present crosslinguistic data on the interaction between partitives and these categories.

6.1 Pseudo-partitives

In addition to the true-partitive relation, partitives are sometimes employed to encode plain quantification as well. For example, the ablative case that is used to encode partitives in Ossetic can encode both: (88) features the true-partitive relation, while (90) denotes plain quantification, i.e., used as a pseudo-partitive (cf. 2.2.4):

(88) Digor Ossetic (Indo-European);

³⁵ I take the Finnish genitive in the singular and nominative in the plural as a non-autonomous accusative case, as the personal pronouns – the only ones to distinguish a dedicated accusative case – show the accusative case here.

42 — Seržant

DE GRUYTER MOUTON

(89) $e=\chi^w esteg-der$ embel-tt-ej jew POSS.3SG=close-COMP friend-PL-ABL(=PART) one 'one of his closest friends' (Avdul Ik'ati. Cardi ajdænæ; courtesy D. Erschler)

(90) Iron Ossetic (Indo-European)

Nartxor-ej er-əffad-toj dəwwe tonne-jə maize- ABL(=PART).SG PRV-grind-PST.3PL two tonne-NUM 'They ground two tons of maize.'
(Bagaev 1965: 156)

Another example is many Arabic varieties in which the separative preposition *min* (and its variants) are required by quantifiers regardless whether or not the meaning is the true-partitive relation or plain quantification, cf. both interpretations in (91):

(91) Gulf Arabic (Afroasiatic)

kathiir min in-niswaan

many from(=PART) DEF-women

'a lot of (the) women'

(Holes 1990: 73)

Yet, 48% of all marked partitives (61/128) in my sample allow for the pseudopartitive function. Note that there is no interaction with macroareas here. The proportion of pseudo-partitives is similar across all macroareas (p = 0.63, χ^2). This means that the co-expression of the true-partitive relation and plain quantification (via pseudo-partitives) is a natural phenomenon. Accordingly, the acquisition of the pseudo-partitive function by partitive constructions originally only denoting the true-partitive relation is a natural drift (Seržant, submitted). What is more, in some languages, partitive constructions are primarily used as pseudo-partitives. Pseudo-partitives may become as pervasive in a language as to become even obligatory in various numeral and quantifier phrases (Greenberg 1978: 301–303), e.g., in Finnish, Russian, Wolof (Greenberg 1978: 287); Rifian Berber (92) or Ossetic (93):³⁶

(92) Central Moroccan or Rif Berber tlata n twrar three GEN hill 'three hills' (Kossmann 2000: 108, 160)

³⁶ I have not found any instances of the true-partitive relation with the genitive in Ossetic and Rifian Berber, which is why these languages attest no possessive strategy in the database.

(93) Ossetic (Indo-European)

fondz xædzar-y

five house-gen.sg

'five houses'

(Arys-Djanaïéva 2004: 107)

Note that the pseudo-partitive function is not equally probable across the syntactic types as can be seen in Table 9.

The stronger preference of the pronominal type to co-express pseudo-partitives might be related to the fact that indefinite quantifiers – from which some of the partitive pronominals emerged – in general tend to retain these two usages, cf. English *some*. The latter has both meanings, the true-partitive usage (*SOME*, stressed) and the quantifier usage (*sm*, unstressed).

Finally, there is also a difference in the probability of the pseudo-partitive function across the different marking strategies of the NP-internal syntactic type, see Table 10. The possessive strategy is most likely to combine true-partitive and pseudo-partitive usages, in contrast to the strategies based on spatial metaphors.

Table 9: Pseudo-partitive usage across different syntactic types (Cramer's V = 0.26, $p < 0.01 \chi^2$).

	NP-internal	Pronominal
Pseudo-partitive	41 % (42)	75% (19)
True partitive only	59% (61)	25% (6)

Table 10: Pseudo-partitive function across strategies (unclear and unmarked strategies excluded).

	Locative	Separative	Possessive
Pseudo-partitive	16% (3)	38% (20)	67 % (16)
True partitive only	84% (16)	62% (32)	33% (8)

6.2 Partitives with intensional and hypothetical predicates

Generalized partitives often interact with intensional and hypothetical predicates. Intensional verbs are verbs that allow for two interpretations of their objects: a specific or transparent meaning (e.g. the speaker has a particular referent in mind) and an opaque, property-denoting, non-referential meaning, i.e. with no existential presupposition (Borschev et al. 2008; Partee 2008; Quine 1960: Section

32; Zimmermann 1993). ³⁷ For example, the English verb *to seek* does not require its object to exist, as one can seek magic items or a new planet, whereas other verbs require their objects to exist under normal circumstances (e.g. *to look at, to destroy*, etc.). Some modern Indo-European languages have even lexicalized the generalized partitive (the possessive strategy) with intensional verbs. For example, the Lithuanian verb *ieškoti* 'to seek' or the verb *iskat*' 'to seek' in Old and dialectal Russian require genitive marking of the object, which goes back to the partitive genitive (Seržant 2014a, 2014b with preceding literature). The same holds for 'want' in Lithuanian:

(94) Lithuanian (Indo-European)

Noriu stal-o / stal-ą su want.prs.1sg table-gen(=part).sg / table-acc.sg with keturi-omis kėd-ėmis four-INS.pl. chair-INS.pl. 'I want (to have) a table with four chairs.' (Ambrazas et al. 2006: 486)

Similar to intensional (and negated (Section 6.3)) predicates are predicates that denote hypothetical events. For example, grammatical categories that are related to future and modality (94) such as the future tense, imperative mood (97) or purposive constructions ((95)–(96)) denote events that are hypothetical and non-referential. These contexts also occasionally allow partitive marking in different languages:

(95) Lithuanian (Indo-European)

Važiuosiu egl-ės pirkti drive.FUT.1SG Christmas_tree-GEN(=PART).SG buy.INF 'I will go (to a marketplace) to buy a Christmas tree.' (Seržant 2014a: 290)

(96) North Russian (Indo-European)

Pošla golovy poloskat'
go.PST.F.SG head.GEN(=PART).SG wash.IPFV.INF
'She went to wash (her) head'.
(Mansikka in Seržant 2014b: 298)

³⁷ Intensional meanings are concepts and are opposed to extensional meanings which have referents; they should not be confused with *intenTional* contexts (Cruse 2000: 21).

CORRECTED PROOF

DE GRUYTER MOUTON

Typology of partitives — 45

(97) North Russian (Indo-European)

Prinesite okutki

bring.PFV.IPV.2PL blanket.GEN(=PART).SG

'Bring the blanket!'

(Mansikka in Seržant 2014b: 299)

A situation similar to North Russian or Lithuanian is found in Latgalian (Nau 2014: 227) and in a number of Finnic languages. In Finnic languages such as Finnish, Karelian, Ingrian or Veps, embedded purpose clauses controlled by a motion verb, modal verbs such as 'want' or 'try' or future auxiliaries such as 'to be going to' may condition the partitive marking on the object NP (Larsson 1983: 84–85, 92–93, 103–104).

Partitives with predicates denoting hypothetical events are also found outside Europe. The partitive prefix *ni*- in Cherokee (Iroquoian) may denote hypothetical events rendered by English 'almost' (*I almost forgot* ...) (Montgomery-Anderson 2008: 313). Similarly, the partitive verbal clitic *-te* in Apma "is often exploited to underline the uncertainty of hypothetical situations, desires, requests and attempts", for example, in irrealis expressions of intention or prospect (Schneider 2010: 167):

(98) Apma (Austronesian; Vanuatu)

Ani na=n veb=te nge teweb.

but 1SG=IRR talk=PART just a.little.bit
'But I'll just talk a little bit.'

(Schneider 2010: 167)

Similarly, the dedicated partitive case (-*DA*) of Yakut and Tofa (Turkic, Siberian) is only used with imperatives (Ubrjatova et al. 1982: 134), and in Dolgan also with intended events ('I will tell you a story(PART)' (Ubrjatova 1985: 117).

The correlation between partitives and hypothetical events is not accidental. It has been observed in the literature that hypothetical events (e.g., irrealis) may be encoded by different kinds of object demotion strategies such as antipassive, for example (Givón 2001: 168). Dahl (1971) and Borschev et al. (2008) observe that the partitive genitive and the subjunctive in Russian have a common semantic core in that both are typically non-presuppositional and non-entailing, the former on the level of the NP's referent and the latter on the clause level (cf. Borschev et al. 2008).

Finally, it is sometimes assumed that the hypothetical meaning (e.g., 'almost' in Cherokee mentioned above) is related to the aspectuality-related function of partitives that entails lack of culmination of a telic event. Thus, the non-culminating meaning 'to shoot at' seems to be semantically close to the meaning 'to almost shoot (to death)' (cf. Section 6.4).

6.3 Partitive under negation

Interaction between partitives and predicate negation is not typologically totally uncommon (Miestamo 2014). Approximately 14% (18/128) of marked partitive expressions in my database show some interaction with predicate negation. Partitives are particularly likely to interact with negation in two areas: Europe and Vanuatu (Arkadiev and Kozhanov forthcoming; Budd 2014; Miestamo 2014; Koptjevskaja-Tamm and Wälchli 2001).

In what follows, I roughly distinguish between two types of interaction between partitives and predicate negation. Both types are diachronically interrelated. The first type (i) pertains to those partitives that have become obligatory in at least some contexts of clausal negation. This type often does not contribute an invariant additional meaning to the sentence interpretation and sometimes is comparable to a discontinuous negation marker (along with the main negation). The second type (ii) is found when the partitive is optional in all contexts of negation. Its primary and invariant function is then emphasis of the negation.

I begin with the first type (i). In Europe, some languages require partitive marking on the object or on the (intransitive) subject of a negated predicate (see Arkadiev and Kozhanov, forthcoming, for an overview). For example, Polish and Lithuanian require the direct object to be marked by the genitive (the possessive strategy) while it is marked by the accusative case in the affirmative clauses, cf.:

(99) Lithuanian (Indo-European)

a. Ne-mačiau tavęs šiandien. b. Mačiau tave šiandien. NEG-see.PST.1SG 2SG.GEN(<PART) today see.PST.1SG 2SG.ACC today 'I didn't see you today.' 'I saw you today.'

The same is true for many Finnic languages with respect to the partitive case and for French with respect to the partitive preposition de (the separative strategy) when the object is indefinite.

It can thus be said that, in these languages, partitives form part of discontinuous negation and are as such highly syntacticized, i.e., blind to any semantic considerations. In some rare instances, even the partitive marking alone may imply negation. For example, Nau (2014: 226) reports that in Latgalian (Baltic, Indo-European) the partitive-genitive marking can alone signal negation if the verb is in the subjunctive (also referred to as *conditional mood*):

(100) Latgalian

kurs ta tī āstu tūs zuoļu who PRT PRT eat.SUBJ DEM.GEN(=PART).PL medicine.GEN(=PART).PL 'who would (want to) take this medicine' (implying 'nobody would take it') (Nau 2014: 227)

Obligatory partitives under negation are found also outside Europe. For example, the partitive marker te(he) in Raga or re in Araki are obligatory under negation (François 2002: 68; Vari-Bogiri 2011: 149). Similarly, the partitive pronoun, turned clitic, becomes obligatory with transitive verbs with non-generic objects in Paamese (Crowley 1982: 147), Lewo, Lamen, South-East Ambrym, Atchin (Early 1994: 81, 84–86, 89) and with prohibitives in Apma (Schneider 2010: 127):

(101) Paamese (Austronesian; Oceania)

Ro-longe-*(tei) inau

3SG.NEG-hear-*(PART) 1SG

'He didn't hear me.'

(Crowley 1982: 145)

The partitive prefix ni- is used in conjunction with the negation marker - $\acute{v}\acute{v}na$ in nominalized subordinate predicates in Cherokee:

(102) Cherokee (Iroquoian; USA)

ni-uu-yóosiisk-vívna a-ali-stáyvvhvska

PART-3-hungry.NEG.NMLZ 3A-MID-fix.a.meal.PRS

'He's eating while he's not hungry.'

(Montgomery-Anderson 2008: 315)

In the following example from Rapa Nui, the genitive preposition seems to be motivated by the negation as well:

(103)Rapa Nui (Austronesian) Kai toe tā'ana 0 te ika, o te 'ura, NEG.PFV remain POSS.3SG.A GEN DET fish GEN DET lobster 0 te kō'iro. GEN DET conger_eel 'There was no fish, lobster or conger eel left for her.' (Kieviet 2017: 254)

Similarly, Thakali requires genitive marking on indefinite pronouns in the S and P position if the predicate is negated (Georg 1996: 83–4):³⁸

(104) Marphatan Thakali (Sino-Tibetan, Bodic) *na-se su-e a mran ju.*1SG-ERG INDEF-GEN NEG see AUX

³⁸ However, since the genitive does not seem to encode partitive relations otherwise in Thakali, it is not entirely clear whether the genitive-under-negation really stems from an original partitive function. For this reason, Thakali has no entry in the database.

'I haven't seen anyone.' (Georg 1996: 84)

Finally, some languages attest the obligatory partitive marking under predicate negation in the earlier stages. For example, the partitive preposition *o* was obligatory with definite objects under negation in Middle Welsh (Borsley et al. 2007: 312). The partitive genitive in early Old Russian also seems to have been obligatory, in contrast to Modern Russian.

I turn now to the second type of interaction between partitives and negation (ii). Many languages employ partitives for expressing emphasis under negation. This is the historical source predating the obligatoriness in type (i) above, as was first argued by Kuryłowicz (1971). Thus, many Bantu languages employ class 16 or 17 partitive (=locative) class markers under predicate negation (Devos and van der Auwera 2013; Persohn and Devos 2017: 20). For example, the partitive (=locative) particle =khwo in Luhya (Bantu) "serves to reinforce negation" (Persohn and Devos 2017: 20). So does the partitive-locative preposition m in Ancient Egyptian (Winand 2015: 539–540). The particle =te in Apma conveys an emphatic meaning in negated transitive sentences in Apma but is obligatory with the existential bibi 'to be' (Schneider 2010: 127, 168–169; cf. also Crowley 1982: 141 on Paamese; Budd 2014: 555–556):

(105) Apma (Austronesian; Vanuatu)

'What did you kill yesterday?'

- a. Na=t=ba ih bamte abma=nga.

 1SG=PFV=NEG hit make.die something=NEG
 'I don't kill things.'

 (Schneider 2010: 169)
- b. Na=t=ba ih bamte=te abma=nga.

 1SG=PFV=NEG hit make.die=PART something=NEG
 'I didn't kill anything.'

While type (i) represents obligatorification of partitives in the context of predicate negation, the second type (ii) seems to predate the first one. The original emphasis under negation was produced by making a stronger claim than what would actually be sufficient in the context, cf.:

- (106) Have you seen the dog with black paws here?
 - a. No, I haven't seen that dog.
 - b. No, I haven't seen any dog here.

(106b) is a stronger statement than (106a) in that it entails the latter but not vice versa. This is due to the reverse entailments under negation: the weaker the

reference, the stronger the statement. If both options are available in the language, the stronger option is typically emphatic in that it provides more information than is actually requested. Generalized partitives are inherently indefinite and may pattern very similar to free-choice quantifiers such as any in (106b).

6.4 Event quantification and contribution to the aspectual interpretation

The ability of partitive-case alternations to affect the aspectuality interpretation of the predicate is, of course, very uncommon (Ö. Dahl in Koptjevskaja-Tamm and Wälchli 2001: 665). However, this is rooted in the fact that fusional case of the archaic Indo-European type is itself typologically rare, rather than in the very semantic mechanism that is actually quite common. Once partitives are defined semantically (9), many more partitives may be observed to interact with aspectuality in a quite uniform way. Thus, 10% of the marked partitives in my sample (Table 11 below) attest functions pertaining to the domain of aspectuality. This does not come as a surprise given that various strategies of P demotion (antipassives, incorporation, etc.) also have been observed to correlate with non-culminating events (Givón 2001: 168).

The aspectual meaning triggered by partitives is typically the *delimitative non-culminating* meaning comparable to English 'somewhat', 'for a while' or 'a bit' used adverbially ("delimitative" in the Slavic tradition, cf. Paducheva 1998; Mehlig 2006, 2007; Seržant 2014b; "partial completion" in Apma in Schneider 2010: 167; often translated as 'a bit' in a number of Bantu languages, Persohn and Devos 2017):

(107) North Russian (Indo-European)

Ja otvorju dverej

1SG open.FUT.1SG door.GEN.PL

'I will somewhat/partly open the door(s).'

(Malyševa in Seržant 2015a: 388)

(108) Finnish (Uralic)

Hän avasi ikkunaa

3SG.NOM opened window.PART.SG

One of the interpretations: '(S)he opened the window (for a while/partly/somewhat).'

(Kiparsky 1998)

Note, however, a slight difference between, on the one hand, North Russian, Russian (Mehlig 2006, 2007; Paducheva 1998; Paykin 2014; Seržant 2014b), Polish

(Wierzbicka 1967), Lithuanian (Seržant 2014a) and Finnish on the other hand. In the former group of languages, the partitive marking of the object triggers only the delimitative interpretation (i.e., to do something for a while and then stop without achieving the culmination) while, in Finnish, the partitive is less committal and expresses any kind of non-culminating accomplishments, including the progressive interpretation ('She was opening the window') alongside the delimitative interpretation ('She somewhat opened the window'). This means that the partitive marking interacts with aspect in the narrow sense in the first group of languages, i.e., it is compatible only with perfective verbs or imperfective iteratives that entail a series of perfective actions. By contrast, the partitive of Finnish only highlights the activity phase of an accomplishment verb, not differentiating between imperfective and perfective non-culminations (Seržant 2014b: 285, 2015a: 371–386).

Similar aspectuality-related functions are also attested outside of Europe. For example, Rapa Nui has the partitive verbal degree modifier '*apa* 'somewhat, kind of' (Kieviet 2017: 340), cf. its argument-related (109) with the aspectuality-related usage (110):

- (109)Rapa Nui (Austronesian) Ko 'apa rova'a mai 'nā te me'e pāreherehe matā. obsidian PRF **PART** obtain hither CONT DET thing piece 'We obtained a few pieces of obsidian.' (Kieviet 2017: 340)
- (110) Rapa Nui (Austronesian)

 Ko 'apa ora 'iti 'ā a au.

 PRF PART live little CONT DET 1SG
 'I am somewhat recovered.'

 (Kieviet 2017: 340)

Many languages of Vanuatu such as Apma, Araki, Bierebo, Erromangan, Lewo and a few others have partitive quantifiers with aspectuality-related functions (Budd 2014: 544–545; Early 1994; Schneider 2010: 167–170).

³⁹ Sometimes, somewhat misleadingly, the range of interpretations of a Finnish predicate is referred to as *imperfective*. This is probably rooted in the terminological confusion in the literature on aspectuality. In this paper, I take the notion of aspect to refer to *viewpoint aspect* with an opposition between perfective and imperfective aspect (*the bidimensional approach to aspectuality* in Bertinetto 1997; Smith 1997; Sasse 2002; just to mention a few). A Finnish VP containing a partitive-marked object may also have a perfective reading of doing something for a while and then stopping before the culmination, cf. (108). Building on Kiparsky (1998). I have argued elsewhere (Seržant 2014b: 285; 2015a: 386) that the partitive case in Finnish concerns marking a VP as culminating vs. non-culminating. It thus pertains to the domain of actionality rather than aspect.

Similarly, the partitive pronoun in Eibela may either quantify the referents of the pronoun, cf. the argument-related interpretation in (111b), or quantify the predicate, cf. the predicate-related interpretation in (111a). Example (111) runs fully parallel to the Finnish textbook example with *ampua* 'to shoot' (*inter alia*, Kiparsky 1998)⁴⁰:

(111) Eibela (Bosavi; Papua New-Guinea)

nɛ:na: i:jɛ: o:-mɛ:na:

1DU 3.PART shoot-FUT.1

(a) 'We two will shoot at them.'

or (b) 'We two will shoot some of them.'

(Aiton 2016:119)

The delimitative meaning is also found in a number of Bantu languages when the locative-partitive class markers attaches to the verb (Persohn 2017a; Persohn and Devos 2017):

(112) Ruund (Atlantic-Congo)

ku-mw-iimikà-p

INF-1-stop-LOC16(=PART)

'to stop him for a minute'

(Persohn and Devos 2017: 17)

Thus, the delimitative meaning sometimes affects not only accomplishment verbs but also some achievement verbs. The partitive particle 'apa in Rapa Nui is also used in constructions that refer to after-effects of a preceding event in a iamitive construction ⁴¹ in combination with the adverb 'already' (Montgomery-Anderson 2008: 313). This is in parallel to the partitive genitive in North Russian and in Eastern Lithuanian dialects where it is used to quantify over the after-effects of an event of transfer (Seržant 2014a: 288–290, 2014b: 286).

The aspectuality-related function is sometimes extended to include two further functions: a modal, conative function 42 and a pragmatic one used as a politeness strategy.

The conative function is found, e.g., in Finnish. Outside of Europe, the conative function is found, for example, in Lewo, cf. (113).

⁴⁰ The achievement verb 'to shoot' is coerced into an accomplishment 'to shoot to death' in which the preparational stage is the very shooting event whereas the inherent endpoint is the moment of striking, cf. Seržant (2015a: 385).

⁴¹ Iamitive (introduced by Olsson 2013 and Dahl and Wälchli 2016: 327–328) is derived from Latin *iam* 'already' and refers to grammaticalized constructions that may be rendered into English with 'already'. The term has been introduced to distinguish these constructions from the semantically related category of perfects.

⁴² Sometimes referred to as cessative, especially in the Finnish tradition.

(113) Lewo (Austronesian; Oceania)

Ne-suma na sineun sape na-kan re kumpui. 1SG-stayed now 1SG.wanted COMP 1SG-eat PART pork

- (a) 'After a while I wanted to eat a bit of pork.'
- (b) 'After a while I wanted **to try** eating some pork.' (Early 1994: 81)

The politeness function derives from the delimitative function on the prediacte ('a little bit', 'somewhat') and from the (pseudo-) partitive function on the object. These functions soften the content expressed by the predicate. For example, the partitive quantifier =ko (the pronominal type) in Oshikwanyama may also be used for politeness. In (114), it softens the hortative meaning and the intended meaning is to offer help by taking over some of the clothes (Halme-Berneking 2017: 147): 43

(114) Oshikwanyama (Atlantic-Congo; Angola, Namibia)

Ha(nd)í-koshé=ko.

1SG.wash-OPT=LOC(=PART)

'Let me wash some (for you).'

(Halme-Berneking 2017: 147)

The politeness function has been reported for Polish (Indo-European), Finnish dialects (Larsson 1983: 75), Hidatsa (Siouan; Park 2012: 481) a number of Bantu languages (Halme-Berneking 2017: 147; Persohn and Devos 2017) such as Fwe (Gunnink 2018: 132, 274).

The syntactic types vary considerably as to their compatibility with the aspectuality-related functions. The NP-internal type in which the partitive marker resides within the argument NP (cf. (107)–(108)), is much less likely to have aspectuality-related functions than the pronominal type (cf. (109)–(114)).

Table 11.	Syntactic types vs	aspectual functions	(statistically significant).44
Table 11:	SVIIIALIIL IVDES VS.	aspectual fullcilons	isialisiicaliv sigiiiiicaiiii.

	NP-internal	Pronominal
Aspectuality-relevant	5% (5)	32% (8)
Irrelevant	95 % (97)	68 % (17)

⁴³ Similar function of the so-called "partitive" quantifier *=po* is found in Nyakyusa (Bantu; Tanzania; Persohn 2017a: 159). However, this marker does not seem to be able to encode the truepartitive relation of the object which is why it was not included in the sample.

⁴⁴ p < 0.005 (Fisher exact test).

Most probably this effect is related to the closer attachment of pronominals to the verb as opposed to NP-internal partitives that are highly integrated within the argument NP. Indeed, the partitive pronominals in a number of Vanuatu languages, for example, may occur immediately next to the verb (Budd 2014: 553–556). Compare *tuut* 'some' in Avava which occurs next to the predicate in (115):

```
115 I-yan tuut emer ki
3SG.R-eat PART eel DEM
'He ate some of the eel.'
```

Somewhat in parallel, the place class markers ku- responsible for the true-partitive interpretation is cross-indexed on the verb by the place marker ko- in (116) (as a kind of *alliterative concord*) in Luvale:

```
(116) Luvale (Bantu)

a-tambwile-ko ku-ma-nyina

1.PST-take-PST-PART(=cl17) PART(=cl17)-6-blood

'He took some of the blood.'

(Persohn and Devos 2017: 22)
```

Many other Bantu languages frequently mark partitives only on the verb, and precisely these languages attest aspectuality-related functions (cf. Persohn and Devos 2017 also with a diachronic explanation).

Indeed, A-quantifiers that scope over the entire predicate generally tend not to be syntactically integrated under some NP (cf. Keenan and Paperno 2012: 948). Contrast the English quantifier *a lot* in its argument-related usage (117a) with the predicate-related usage in (117b):

(117) a. He bought **a lot** of flowers vs. b. He has been buying flowers **a lot**

6.5 Synopsis of the additional functions

I have illustrated above that generalized partitives tend to have additional functions alongside encoding the true-partitive relation. They may pattern as pseudopartitives and may interact with predicate categories such as negation or aspectuality. These additional functions are, however, not equally frequent in the world's languages (Table 12).

The morphological homonymy between true-partitives and pseudo-partitives (48%) is a very frequent phenomenon that has to do with the universal tendency

54 — Seržant

DE GRUYTER MOUTON

Table 12: The frequency of related functions in the database (unmarked partitives excluded). 45

True-partitive	Pseudo-partitive	Negation	Aspectuality
100%	48% (61/128)	14% (18/128)	10% (13/128)

towards coding syncretism of two different meanings: the true-partitive relation and plain quantification (see Section 7 below).

Another important result emerges from the figures in Table 12. The additional functions form an implicational scale such that only those partitives that may pattern as pseudo-partitives can have functions pertaining to predicate negation and/or aspectuality:

(118) Functional scale of partitives

Pseudo-partitives > Negation / Aspectuality

This is also supported by the fact – not visible from Table 12 – that partitives with functions pertaining to negation and/or aspectuality form a subset of the partitives that may pattern as pseudo-partitives in the database. Thus, semantic properties that are recruited during the grammaticalization of partitives into markers (co-) expressing predicate negation or aspectuality are properties of plain quantification found in pseudo-partitives, and not of the true-partitive relation.

Pseudo-partitives serve to denote a particular – often indefinite – quantity of a particular kind. Accordingly, pseudo-partitives have a similar effect on event quantification as other indefinite quantifiers like *somewhat* or *a little bit* in English. In parallel to English a *little bit*, pseudo-partitives acquire the ability to function as event measurers (*to work a little bit*) parallel to the nominal domain where they are object-measurers (*a little bit of tea*).

There is another piece of evidence for the claim that it is precisely the pseudo-partitive – and not the true-partitive – usage that leads to the manipulation of the aspectual interpretation of the verb and/or to the interaction with predicate negation. Markers that only have the indefinite-quantity meaning but cannot encode the true-partitive relation are also found to yield exactly the same, delimitative meaning on the predicate, while they do not have the meaning of the true-partitive relation at all. For example, the marker -woh- – not a partitive

⁴⁵ I have excluded the unmarked partitives (like lit. 'some us') because these are inherently indistinguishable from simple, non-proportional quantification (like *some people*). This leaves a total of 128. The difference between the pseudo-partitives and negation is statistically significant (p < 0.001) and the difference between negation and aspect is obviously not.

according to the definition (9) adopted in this study, but glossed as partitive in Evans (2003) – yields the delimitative meaning 'a bit, for a while' in Bininj Gun-Wok:

(119) Bininj Gun-Wok (Gunwinyguan; Australia)
nga-woh-dirri
1-QUANTIFIER-play.NONPST
'I'll play for a while.'
(Evans 2003: 501)

The same applies to negation. Certain indefinite quantifiers that do not attest the true-partitive meaning are found to yield emphasis of negation. For example, the particle -woh- in the context of negation in Bininj Gun-Wok also yields emphasis (Evans 2003: 502). Similarly, the particle =po in Nyakyusa (Bantu; Tanzania) "strengthens the negation, yielding a meaning along the lines of 'not a bit' or 'no at all'" (Persohn 2017a: 158). To the best of my knowledge, neither of these markers is found with the true-partitive relation. Moreover, the locative class marker -po in Nyakyusa (in contrast to the partitive locatives in most other Bantu languages) has never been a true-partitive marker - it indirectly stems from a quantifier 'a little bit'. ⁴⁶

7 A universal trend toward syncretism of the truepartitive relation and plain quantification

In this section, I argue that there is a universal trend toward syncretism of two distinct meanings: the true-partitive relation and the plain quantification.

Above (Section 6.5), I have argued that 48% of all *marked* partitives in my database allow for the pseudo-partitive usage (Table 12), i.e., 48% of all marked partitives do not morphologically differentiate between the two meanings. This figure excludes all partitives encoded by the unmarked strategy. Yet, this strategy is inherently ambiguous between the true-partitive relation and plain quantification. Moreover, this is also one of the most frequent strategies (ca. 25% of all partitives in my database, cf. Table 4, Section 4.3.2 above). If one also adds the unmarked strategy, then the frequency of co-expressing these two meanings becomes even larger, i.e. 61% (104/171). Finally, implicit expressions are still another instance of the coding overlap between plain quantification and the true-partitive relation. Recall that implicit expressions combine the constructional means

⁴⁶ Originally, it was conditioned by the locative-class noun *panandt* meaning 'a little' (*pa-nandt* '16-little'), which was often dropped and the clitic *-po* alone started triggering the meaning 'a little bit' (Persohn 2017a: 157–158, 2017b: 68–69; Persohn and Devos 2017: 24).

regularly used for plain quantification with contextual inferences to yield the truepartitive interpretation (Section 2.2.2). Unfortunately, I do not have quantitative data on implicit expressions but, impressionistically, they seem to be very frequent crosslinguistically.

From this it follows that at least occasional resorting to the constructional means of plain quantification when encoding the true-partitive relation is a default, in fact, universal option. This option is available to different degrees across languages, depending on such factors as: the properties of the superset (e.g. less frequently with pronominal supersets) and on the presence of a fully grammaticalized partitive construction in the language.

I claim that the reason for this universal trend toward syncretism is that the true-partitive relation (e.g., *some of the students*) provides an "extra bit" of information with the superset (*the students*) that is rarely essential and, therefore, often need not be discriminated from the meaning of plain quantification (*some students*) at all. The relation of a referent to some superset (*anchoring*) is often irrelevant in the discourse and, even if not, may often be inferred from the context anyway. Indeed, the true-partitive relation is only very rarely explicitly coded in discourse even in languages which do have marked partitive constructions to encode the true-partitive relation. Thus, a small corpus survey of the Russian indefinite quantifier *neskol'ko* 'some' reveals that the probability of it occurring in the construction that explicitly encodes the true-partitive relation (based on the preposition *iz* 'from') is extremely low: ca. 1.6% of all its occurrences. This is despite the fact that this quantifier is among the most frequent ones to occur in this construction (cf. Table 1 above).⁴⁷

8 Conclusions

In this paper, I have demonstrated that partitives take different shapes both in terms of syntactic structure and morphology. As for the former, there is a cline from – possibly less grammaticalized – partitives that do not require adjacency and form no constituency between the quantifier and the restrictor NPs, to partitives with rigid head-dependency relations with an adjacency requirement and, finally, to partitives in which both merge into a single prosodic unit. Since few grammars

⁴⁷ Among the first hits, excluding all irrelevant uses (such as the use as an adverb quantifier 'somewhat'), there were only two instances (1.6%) of it occurring in the subset of a partitive (marked with *iz* 'from') as opposed to 120 of simple quantification *neskol'ko čelovek* 'some people' (based on the Spoken subcorpus of the Russian National Corpus).

provide information on the syntactic properties of partitives, I could not create a statistically representative sample for these types, leaving this for future research.

Formally, there are different ways in how partitives may be coded (if at all). First, there are two syntactic types: The morphological marker may either be an NP-internal marker – most typically an adposition – or a partitive pronominal comprising partitive pronouns, quantifiers and particles. Furthermore, there are four main marking strategies: the unmarked, the possessive, the locative and the separative strategies, which may be further subdivided into more subtypes. The unmarked strategy yields the third syntactic type – juxaposition.

None of the combination of these types and strategies is universal. However, these marking strategies are not randomly distributed across the globe. I have argued that there are strong areal biases in their geographical distribution. First, Eurasia stands out with a clear bias towards NP-internal marking of partitives, dispreferring the unmarked strategy. Secondly, Eurasia strongly prefers the separative strategy among the NP-internal strategies. By contrast, Africa seems to prefer the locative strategy. Many Bantu languages attest this strategy (cf. Persohn and Devos 2017) in addition to a few Songhay, Dogon and Afroasiatic languages of the database.

Finally, I have examined additional functional facets of partitives alongside the true-partitive relation. Thus, partitives may also function as pseudo-partitives, (co-)express negation, hypothetical events and aspectuality. The compatibility with negation and aspectuality-related functions of partitives follows the cline in (118) repeated here as (120) for convenience:

(120) Functional scale of partitives

Pseudo-partitives > Negation / Aspectuality

The synchronic distribution of these functions in my sample suggests that negation and aspectuality are closely related to pseudo-partitives. This distribution also has a diachronic dimension (Seržant, submitted).

I have argued that partitives (co-)expressing aspectuality or predicate negation are generally not so infrequent. However, specifically NP-internal partitives are quite atypical with these functions. Thus, nearly one third of partitive pronominals attest these functions while less than one tenth of NP-internal partitives are found with these functions in my database. From this it follows that the Finnish partitive is not so unusual semantically in the crosslinguistic perspective in having aspectuality and negation-related functions. It is rather unusual in its combination of the locus of realization (NP-internally) and the versatility of the additional, clause-level functions (aspectuality, negation).

Finally, I have also made a universal claim. Even though many languages develop morphological devices for marking the true-partitive relation, there is a

universal preference for at least some degree of syncretism between the truepartitive relation and plain quantification. This preference holds with regard to both type and token frequency. Moreover, the syncretism is also crosslinguistically bidirectional. Thus, there is a universal trend for partitives to resort to the constructional means of plain quantification in terms of the unmarked strategy and implicit expressions (Section 7). And, vice versa, marked partitive constructions also tend to allow the encoding of plain quantification via pseudo-partitives (Section 6.1).

The reason for this universal trend towards morphologically shorter and syntactically simpler constructions, which are primarily used to encode plain quantification, may be explained in terms of coding efficiency. True-partitives provide an "extra bit" of information on the anchoring of the (subset) referent in some superset. However, an explicit anchoring is rarely needed because either the referent is definite anyway and/or the anchoring information can easily be guessed from the discourse. Only rarely must the relation to a superset be emphasized explicitly as the small corpus study from Russian clearly shows. In efficient communication, the meanings and functions that can anyway be derived contextually or that can be guessed on the basis of experience (frequency-based expectations) and/or previous discourse tend to be left unexpressed in order to save the articulatory and processing effort of the hearer and the speaker (Haspelmath 2008a, 2008b, 2008c).

Acknowledgements: This research has been carried out at Leipzig University, project "Grammatical Universals". The support of the European Research Council (ERC Advanced Grant 670985) is gratefully acknowledged. Personally, I cordially thank HH for help with access to materials. Without this archive this work wouldn't have been possible. Søren Wichmann helped me a lot with finding the grammars with sections on partitives. I also thank Peter Arkadiev, Oleg Belyaev, David Erschler, Martin Haspelmath, Nicole Hober and Kurt Malcher. I, furthermore, thank all the anonymous reviewers of the previous versions of this paper. Last but not least, I also thank the editors of Linguistics for their very valuable work notwithstanding the pandemic situation.

Abbreviations (in addition to Leipzig Glossing Rules)

CF counterfactual coll collective

comment marker (in a topic-comment structure)

cons concessive

DISTR distributive
EMPH emphatic
EVT eventual
EXC exclusive
GL goal

iis incomplete intransitive status

INS instrumental
MID middle voice
NF non-finite
NPL non-plural
NONPST non-past
NRLD non-realized
OBLIG obligative

POL politeness marker

POSS possessive

PRSP prospective aspect
REF reflexive/reciprocal

REL relative
TRN transitive
ST stative
SUBEL subelative case

SVN subjective verbal noun

References

Abraham, P. T. 1985. *Apatani grammar (Central Institute of Indian Languages: Grammar Series 12)*.

Mysore: Central Institute of Indian Languages.

Aiton, Grant. 2016. *Grammatical relations and information structure in Eibela: A typological perspective*. Douglas: James Cook University dissertation.

Ambrazas, V., K. Garšva, A. Girdenis, E. Jakaitienė, P. Kniūkšta, S. Krinickaitė, V. Labutis & A. Laigonaitė. 2006. *Dabartinės lietuvių kalbos gramatika*. *A grammar of modern Lithuanian*. Vilnius: Lietuvių kalbos institutas.

Anderson, Gregory D. & David K. Harrison. 1999. *Tyvan.* (Languages of the World/Materials 257). Munich: Lincom.

Arkadiev, Peter & Kirill Kozhanov. forthcoming. Object partitive of negation: An areal perspective. In Björn Wiemer & Arkadiev Peter (eds.), *Convergence and divergence in the eastern Circum-Baltic Area (A triangulation approach)*. Berlin & Boston: De Gruyter Mouton.

Arys-Djanaïéva, Lora. 2004. Parlons Ossète. (Collection Parlons.). Paris: L'Harmattan.

Asher, Ronald E. 1985. Tamil (Lingua Descriptive Studies, 7). London: Croom Helm.

Asher, Ronald E. & T. C. Kumari. 1997. *Malayalam. (Descriptive grammars series, descriptive grammars)*. London & New York: Routledge.

Ashton, Ethel Ostell, E. M. K. Mulira, E. G. M. Ndawula & A. N. Tucker. 1954. *A Luganda grammar*. London & New York: Longmans, Green and Company.

Atoyebi, Joseph Dele. 2010. A reference grammar of Oko: A West Benue-Congo language of North-Central Nigeria. Cologne: Rüdiger Köppe.

- Authier, Gilles. 2009. Grammaire kryz: Langue caucasique d'Azerbaïdjan, dialecte d'Alik. Paris: Peeters.
- Bagaev, Nikolaj K. 1965. Sovremennyj osetinskij jazyk. Čast' 1 (fonetika i morfologija) [Modern Ossetic. Part 1: Phonetics and Morphology]. Ordžonikidze: Severo-Osetinskoe knižnoe izdateľ stvo.
- Barbour, Julie. 2012. A grammar of Neverver. (Mouton Grammar Library 60). Berlin & Boston: De Gruyter Mouton.
- Barlow, Russell. 2018. *A grammar of Ulwa*. Honolulu: University of Hawai'i at Mānoa dissertation. Barwise, Jon & Robin Cooper. 1981. Generalized quantifiers and natural language. *Linguistics and Philosophy* 4. 159–219.
- Bauer, Winifred. 1993. Maori. (Descriptive grammars series). London & New York: Routledge.
- Beller, Richard & Patricia Beller. 1979. Huasteca Nahuatl. In Ronald W. Langacker (ed.), *Studies in Uto-Aztecan grammar 2: Modern Aztec grammatical sketches (Summer Institute of Linguistics Publications in Linguistics 56)*, 199–306. Dallas, TX: Summer Institute of Linguistics and the University of Texas at Arlington.
- Bertinetto, Pier Marco. 1997. Il dominio tempo-aspettuale: Demarcazioni, intersezioni, contrasti. Torino: Rosenberg and Sellier.
- Besnier, Niko. 2000. *Tuvaluan: A Polynesian language of the Central Pacific. (Descriptive grammars series)*. London & New York: Routledge.
- Bhatia, Tej K. 1993. *Punjabi: A cognitive-descriptive grammar. (Descriptive grammars series)*. London & New York: Routledge.
- Bicevskis, Katie, Henry Davis & Lisa Matthewson. 2017. Quantification in Gitksan. In Denis Paperno & Edward Keenan (eds.), *Handbook of quantifiers in natural language*, 281–382. Springer.
- Bodrogligeti, András J. E. 2003. Modern literary Uzbek: A manual for intensive elementary, intermediate, and advanced courses. (LINCOM Language Coursebooks 10). Munich: Lincom.
- Borschev, Vladimir, Elena V. Paducheva, Barbara H. Partee, Yakov Testelets & Igor Yanovich. 2008. Russian genitives, non-referentiality, and the property-type hypothesis. In Andrei Antonenko, John F. Bailyn & Christina Bethin (eds.), *Formal approaches to Slavic linguistics: The Stony Brook Meeting 2007*, vol. 16, 48–67. Ann Arbor, MI: Michigan Slavic Publications.
- Borsley, Robert D., Maggie Tallerman & David Willis. 2007. *The syntax of Welsh. (Cambridge syntax guides*). Cambridge: Cambridge University Press.
- Brownie, John & Marjo Brownie. 2007. Mussau grammar essentials. (Data papers on Papua New Guinea languages 52). Ukarumpa: Summer Institute of Linguistics.
- Budd, Peter. 2014. Partitives in Oceanic languages. In Silvia Luraghi & Tuomas Huumo (eds.), Partitive cases and related categories. (Empirical approaches to language typology 54), 523–562. Berlin & Boston: De Gruyter Mouton.
- Carlier, Anna & Béatrice Lamiroy. 2014. The grammaticalization of the prepositional partitive in Romance. In Silvia Luraghi & Tuomas Huumo (eds.), *Partitive cases and related categories.* (Empirical approaches to language typology 54), 477–520. Berlin & Boston: De Gruyter Mouton.
- Carlson, Robert. 1994. A grammar of Supyire. (Mouton grammar library 14). Berlin & New York: Mouton de Gruyter.
- Casad, Eugene H. 1984. Cora. In Ronald W. Langacker (ed.), Studies in Uto-Aztecan grammar 4: Southern Uto-Aztecan grammatical sketches, 153-459. Dallas: Summer Institute of Linguistics and the University of Texas at Arlington.

- Comrie, Bernard. 1981. Language universals and linguistic typology: Syntax and morphology. Chicago, IL: University of Chicago Press.
- Crippen, James. 2012. The basics of Tlingit verbal structure. http://tlingitlanguage.com/wp-content/uploads/2015/01/Basics-of-Tlingit-Verbal-Structure.pdf.
- Croft, William. 1988. Agreement vs. case marking and direct objects. In Michael Barlow, & Charles A. Ferguson (eds.), *Agreement in natural language: Approaches, theories, descriptions*, 159–180. Stanford, CA: CSLI.
- Crowley, Terry. 1982. *The Paamese language of Vanuatu. (Pacific linguistics: Series B 87)*.

 Canberra: Research School of Pacific and Asian Studies. Australian National University.
- Cruse, David Alan. 2000. Meaning in language. An introduction to semantics and pragmatics.
 Oxford: Oxford University Press.
- Dahl, Östen. 1971. The genitive and the subjunctive in Russian. *Scando-Slavica* 17(23). 135–137.
- Dahl, Östen. 2000. Egophoricity in discourse and syntax. Functions of Language 7. 39-77.
- Dahl, Östen & Bernhard Wälchli. 2016. Perfects and iamitives: Two gram types in one grammatical space. *Letras de Hoje* 51(3). 325–348.
- Davey, Richard J. 2016. Coastal Dhofari Arabic: A sketch grammar. Leiden: Brill.
- Daudey, Henriëtte. 2014. A grammar of Wadu Pumi. Melbourne: LaTrobe University dissertation.
- Davies, William D. 2010. A grammar of Madurese. (Mouton grammar library 50). Berlin & New York: De Gruyter Mouton.
- Devos, Maud, Misago Manoah-Joël & Koen Bostoen. 2017. A corpus-based description of locative and non-locative reference in Kirundi locative enclitics. *Africana Linquistica* 23. 47–83.
- Devos, Maud & Johan van der Auwera. 2013. Jespersen Cycles in Bantu: Double and triple negation. *Journal of African Languages and Linguistics* 43(2). 205–274.
- Dombrowsky-Hahn, Klaudia. 2015. A grammar of Syer (Western Karaboro, Senufo. (Gur Monographs 12). Cologne: Köppe.
- Dowty, David & Belinda Brodie. 1984. The semantics of "floated" quantifiers in a transformationless grammar. In *Proceedings of the Third West Coast Conference on Formal Linquistics (WCCFL 3*), 75–90. Stanford, CA: CSLI Publications.
- Dum-Tragut, Jasmine. 2009. Armenian: Modern Eastern Armenian. Amsterdam: John Benjamins.
- Early, Robert J. 1994. *A grammar of Lewo, Vanuatu*. Canberra: Australian National University dissertation.
- Enç, Mürvet. 1991. The semantics of specificity. Linguistic Inquiry 22. 1–25.
- Enrico, John. 2003. *Haida syntax (Studies in the anthropology of North American Indians*), vol. 2. Lincoln: University of Nebraska Press.
- Evans, Nicholas. 2003. Bininj Gun-Wok: A Pan-dialectal grammar of Mayali, Kunwinjku and Kune. (Pacific Linguistics 541), vol. 2. Canberra: Research School of Pacific and Asian Studies, Australian National University.
- Erdal, Marcel. 2004. A grammar of Old Turkic. Leiden & Boston: Brill.
- Filip, Hana. 1999. Aspect, eventuality types and noun phrase semantics. New York: Routledge.
- Foris, David P. 2000. A grammar of Sochiapan Chinantec. (Summer Institute of Linguistics: Publications in Linguistics 135). Studies in Chinantec Languages 6. Dallas, TX: The Summer Institute of Linguistics and the University of Texas at Arlington.
- Forker, Diana. 2013. A grammar of Hinuq. (Mouton Grammar Library 63). Berlin & Boston: De Gruyter Mouton.
- Frajzyngier, Zygmunt. 2002. A grammar of Hdi. (Mouton Grammar Library 21). Berlin & New York: Mouton de Gruyter.

- François, Alexandre. 2002. Araki: A disappearing language of Vanuatu. (Pacific linguistics 522).

 Canberra: Research School of Pacific and Asian Studies, Australian National University.
- Georg, Stefan. 1996. Marphatan Thakali: Untersuchungen zur Sprache des Dorfes Marpha im Oberen Kāli-Ganṇḍaki-Tal/Nepal. (LINCOM Studies in Asian Linguistics 2). Munich: Lincom.
- Givón, Talmy. 2001. Syntax, vol. 2. Amsterdam & Philadelphia: John Benjamins.
- Givón, Talmy. 2011. *Ute reference grammar. (Culture and language use 3)*. Amsterdam & Philadelphia: John Benjamins.
- Glaser, Elvira. 1992. Umbau partitiver Strukturen in der Geschichte des Deutschen. Sprachwissenschaft 17(2). 113–132.
- Greenberg, Joseph H. 1978. Generalizations about numeral systems. In Joseph H. Greenberg, Charles A. Ferguson & Edith A. Moravcsik (eds.), *Universals of human language*, vol. 3: Word structure, 249–295. Stanford, CA: Stanford University Press.
- Gunnink, Hilde. 2018. A grammar of Fwe: A Bantu language of Zambia and Namibia. Ghent: University of Ghent dissertation.
- Halme-Berneking, Riikka. 2017. Oshikwanyama locative enclitics (and their non-locative functions). *Africana Linquistica* 23. 137–150.
- Haspelmath, Martin. 1993. A grammar of Lezgian. (Mouton grammar library 9). Berlin & New York: Mouton de Gruyter.
- Haspelmath, Martin. 2008a. Frequency vs. iconicity in explaining grammatical asymmetries. *Cognitive Linguistics* 19(1). 1–33.
- Haspelmath, Martin. 2008b. Creating economical patterns in language change. In Jeff Good (ed.), Linguistic universals and language change, 185–214. Oxford: Oxford University Press.
- Haspelmath, Martin. 2008c. A frequentist explanation of some universals of reflexive marking. *Linquistic Discovery* 6(1). 40–63.
- Haspelmath, Martin. 2018. How comparative concepts and descriptive linguistic categories are different. In Daniël Van Olmen, Tanja Mortelmans & Brisard Frank (eds.), *Aspects of linguistic variation*, 83–114. Berlin: DeGruyter.
- Haude, Katharina. 2006. A grammar of Movima. Nijmegen: Radboud Universiteit Nijmegen dissertation.
- Heath, Jeffrey. 1999. A grammar of Koyra Chiini: The Songhay of Timbuktu (Mouton grammar library 19). Berlin: DeGruyter.
- Heath, Jeffrey. 2007. Grammar of Humburi Senni (Songhay of Hombori, Mali). Ms.
- Heath, Jeffrey. 2008. A grammar of Jamsay (Mouton grammar library 45). Berlin: DeGruyter.
- Heath, Jeffrey. 2015. A grammar of Togo Kan. Available at: https://deepblue.lib.umich.edu/bitstream/handle/2027.42/123066/A%20grammar%20of%20Togo%20Kan%20downsized.pdf?sequence=4&isAllowed=y.
- Heath, Jeffrey. 2017. A grammar of Jalkunan (Mande, Burkina Faso). Available at: https://deepblue. lib.umich.edu/bitstream/handle/2027.42/139025/Jalkunan%20grammar%202017% 20reduced.pdf?sequence=1&isAllowed=y.
- Helasvuo, Marja. 1996. A discourse perspective on the grammaticization of the partitive case in Finnish. In Timo Haukioja, Helasvuo Marja-Liisa & Elise Kärkkäinen (eds.), *SKY yearbook of the linguistic association of Finland* 18, 7–35. Helsinki: The Linguistic Association of Finland.
- von Heusinger, Klaus. 2002. Specificity and definiteness in sentence and discourse structure. *Journal of Semantics* 19(3). 245–274.
- von Heusinger, Klaus & Jaklin Kornfilt. 2017. Partitivity and case marking in Turkish and related languages. *Glossa* 2(1). 1–40.

- Hill, Jane H. & Kenneth C. Hill. 1986. Speaking Mexicano: Dynamics of syncretic language in Central Mexico. Tucson, AZ: University of Arizona Press.
- Hober, Nicole. 2020. On the intrusion of the Spanish preposition de into the languages of Mexico. Journal of Language Contact 12(3). 660–706.
- Hoeksema, Jacob. 1996a. Introduction. In Hoeksema Jacob (ed.), *Partitives: Studies on the syntax* and semantics of partitive and related constructions (Groningen-Amsterdam Studies in Semantics 14), 2–24. Berlin & New York: Mouton de Gruyter.
- Hoeksema, Jacob (ed.). 1996b. Partitives: Studies on the syntax and semantics of partitive and related constructions. (Groningen-Amsterdam Studies in Semantics 14). Berlin & New York: Mouton de Gruyter.
- Hofling, Charles A. 2000. Itzaj Maya grammar. Salt Lake City, UT: University of Utah Press.
- Holes, Clive. 1990. *Gulf Arabic (Croom Helm Descriptive Grammars)*. London & New York: Routledge.
- de Hoop, Helen. 1997. A semantic reanalysis of the partitive constraint. *Lingua* 103 2/3. 151–174.
- de Hoop, Helen. 2003. Partitivity. In Lisa Lai Shen Cheng & Rint Sybesma (eds.), *The second Glot International state-of-the-article book*, 179–212. Berlin & New York: Mouton de Gruyter.
- Hualde, José I. & Jon Ortiz de Urbina (eds.). 2003. *A grammar of Basque (Mouton Grammar Library 26*). Berlin & New York: Mouton de Gruyter.
- Huumo, Tuomas. 2010. Nominal aspect, quantity, and time: The case of the Finnish object. *Journal of Linguistics* 46. 83–125.
- Huumo, Tuomas. 2018. The partitive A: On uses of the Finnish partitive subject in transitive clauses. In Ilja A. Seržant & Alena Witzlack-Makarevich (eds.), *Diachrony of differential argument marking (Studies in diversity linguistics*), 383–411. Berlin: Language Science Press.
- Ihsane, Tabea. 2013. EN pronominalisation in French and the structure of nominal expressions. Syntax 16(3). 217–249.
- Ionin, Tania, Ora Matushansky & Eddy G. Ruys. 2006. Parts of speech: Towards a unified semantics for partitives. In Christopher Davis, Deal Amy & Youri Zabbal (eds.), Proceedings of NELS, vol. 36, 357–370. Amherst, MA: GLSA University of Massachusetts.
- Jackendoff, Ray. 1977. X-bar syntax: A study of phrase structure. Cambridge, MA: MIT Press.
- Janssen, Dirk, Balthasar Bickel & Fernando Zúñiga. 2006. Randomization tests in language typology. *Linguistic Typology* 10(3). 419–440.
- Julian, Charles. 2010. A history of the Iroquoian languages. Winnipeg: University of Manitoba dissertation.
- Kari, Ethelbert E. 2004. A reference grammar of Degema. (Grammatische Analysen afrikanischer Sprachen 22). Cologne: Rüdiger Köppe Verlag.
- Keenan, Edward L. & Denis Paperno. 2012. Overview. In Edward L. Keenan & Denis Paperno (eds.), Handbook of quantifiers in natural language, 941–950. Dordrecht: Springer.
- Khalilova, Zaira. 2009. A grammar of Khwarshi. Leiden: Rijksuniversiteit te Leiden dissertation.
- Kibrik, Aleksandr E. 1977. Opyt strukturnogo opisanija arčinskogo jazyka [An attempt at a structural description of Archi] (Publikacii otdelenija strukturnoj i prikladnoj lingvistiki 11–13). Moscow: Izd. Moskovskogo Univ.
- Kieviet, Paulus. 2017. A grammar of Rapa Nui. Berlin: Language Science Press.
- Kim, Alan Hyun-Oak. 1995. Word order at the noun phrase level in Japanese: Quantifier constructions and discourse functions. In Pamela Downing & Michael P. Noonan (eds.), Word order in discourse, 199–246. Amsterdam & Philadelphia: John Benjamins.

- Kiparsky, Paul. 1998. Partitive case and aspect. In Miriam Butt & Wilhelm Geuder (eds.), *The projection of arguments: lexical and compositional factors*, 265–307. Stanford, CA: CSLI Publications.
- Koptjevskaja-Tamm, Maria. 2001. A piece of the cake and a cup of tea: Partitive and pseudopartitive nominal constructions in the Circum-Baltic Languages. In Östen Dahl & Maria Koptjevskaja-Tamm (eds.), *The Circum-Baltic languages. Typology and contact*, vol. 2: *Grammar and typology* (Studies in language companion series), 523–568. Amsterdam & Philadelphia: John Benjamins.
- Koptjevskaja-Tamm, Maria. 2009. A lot of grammar with a good portion of lexicon: Towards a typology of partitive and pseudopartitive nominal constructions. In Johannes Helmbrecht, Yoko Nishina, Yong-Min Shin, Stavros Skopeteas & Elisabeth Verhoeven (eds.), Form and function in language research: Papers in honour of Christian Lehmann, 329–346. Berlin & New York: Mouton de Gruyter.
- Koptjevskaja-Tamm, Maria & Bernhard Wälchli. 2001. The Circum-Baltic languages: An areal-typological approach. In Östen Dahl & Maria Koptjevskaja-Tamm (eds.), *The Circum-Baltic languages: Typology and contact*, vol. 2: Grammar and typology, 615–750. Amsterdam & Philadelphia: John Benjamins.
- Kossmann, Maarten G. 2000. Esquisse grammaticale du rifain oriental (Société d'études linguistiques et anthropologiques de France 387). Leuven: Peeters.
- Krifka, Manfred. 1992. Thematic relations as links between nominal reference and temporal constitution. In Ivan A. Sag & Szabolcsi Anna (eds.), *Lexical matters*, 29–53. Stanford, CA: CSLI.
- Kuryłowicz, Jerzy. 1971. Słowiański genetivus po negacij [The Slavic genitive after negation]. In Stanisław Urbańczyk (ed.), *Sesja naukowa międzynarodowej komisji budowy gramatycznej języków słowiańskich*, 11–14. Wroclaw: Zaklad Narodowy im. Ossolinskich.
- Ladusaw, William A. 1982. Semantic constraints on the English partitive construction. In Daniel P. Flickinger, Marlys Macken & Nancy Wiegand (eds.), Proceedings of the First West Coast Conference on Formal Linguistics (WCCFL 1), 231–242. Stanford, CA: Stanford Linguistics Association.
- Lamberti, Marcello & Roberto Sottile. 1997. *The Wolaytta language (Studia linguarum Africae orientalis 6)*. Cologne: Rüdiger Köppe.
- LaPolla, Randy J. & Chenglong Huang. 2003. A grammar of Qiang with annotated texts and glossary (Mouton grammar library 31). Berlin & New York: Mouton de Gruyter.
- Larsson, Lars-Gunnar. 1983. Studien zum Partitivgebrauch in den ostseefinnischen Sprachen. Acta Universitatis Upsaliensis (Studia Uralica et Altaica Upsaliensia 15). Uppsala: Borgströms Tryckeri AB.
- Leer, Jeffry A. 1991. The schetic categories of the Tlingit verb. Chicago, IL: University of Chicago dissertation.
- Levin, Beth. 1993. *English verb classes and alternations: A preliminary investigation*. Chicago, IL: University of Chicago Press.
- Luraghi, Silvia & Seppo Kittilä. 2014. Typology and diachrony of partitive case markers. In Silvia Luraghi & Tuomas Huumo (eds.), *Partitive cases and related categories (Empirical approaches to language typology 54*), 17–62. Berlin & Boston: De Gruyter Mouton.
- Lytkin, V. I., K. Je. Majtinskaja, K. Redei, J. Guja, A. P. Feoktistov & G. I. Jermuškin. 1975. Osnovy finno-ugorskogo jazykoznanija. Pribaltijsko-finnskie, saamskij i mordovskie jazyki. Moscow: Nauka. [Foundations of the Finno-Ugric linguistics. Finnic, Saamic and Mordvin languages].

- Mehlig, Hans Robert. 2006. Glagol'nyj vid i vtoričnaja gomogenizacija oboznačaemoj situacii: K upotrebleniju delimitativnogo sposoba dejstvija v russkom jazyke [Verbal aspect and the secondary homogenization of the encoded situation: About the use of the delimitative aktionsart in Russian]. In Folkmar Leman (ed.), Semantika i struktura slavjanskogo vida 4, 235–276. Munich: Sagner.
- Mehlig, Hans Robert. 2007. Aspect and bounded quantity complements in Russian. In Susan Rothstein (ed.), *Theoretical and crosslinguistic approaches to the semantics of aspect (Linguistics Today 110*), 257–290. Amsterdam & Philadelphia: John Benjamins.
- Miestamo, Matti. 2014. Partitives and negation: A cross-linguistic survey. In Silvia Luraghi & Tuomas Huumo (eds.), Partitive cases and related categories (Empirical approaches to language typology 54), 63–86. Berlin & Boston: De Gruyter Mouton.
- Montgomery-Anderson, Brad. 2008. *A reference grammar of Oklahoma Cherokee*. Lawrence, KS: University of Kansas dissertation.
- Mosel, Ulrike & Even Hovdhaugen. 1992. Samoan reference grammar. Oslo: Scandinavian University Press.
- Nau, Nicole. 2014. Differential object marking in Latgalian. In Axel Holvoet & Nau Nicole (eds.), Grammatical relations and their non-canonical encoding in Baltic (Valency, Argument Realization and Grammatical Relations in Baltic 1), 207–255. Amsterdam & Philadelphia: John Benjamins.
- Næss, Åshild. 2004. What markedness marks: The markedness problem with direct objects. Lingua 114. 1186–1212.
- Næss, Åshild. 2007. Prototypical transitivity. Amsterdam & Philadelphia: John Benjamins.
- Nedyalkov, Igor. 1994. Evenki. In Peter Kahrel & René van den Berg (eds.), *Typological studies in negation (Typological studies in language 29*), 1–34. Amsterdam & Philadelphia: John Benjamins.
- Nevskaya, Irina. 2017. Innovations and archaisms in Siberian Turkic spatial case paradigms: A Transeurasian historical and areal perspective. In Martine Robbeets & Bisang Walter (eds.), Paradigm change in the Transeurasian languages and beyond, 257–286. Amsterdam & Philadelphia: John Benjamins.
- Olsson, Bruno. 2013. *Iamitives: Perfects in Southeast Asia and beyond*. MA thesis. Stockholm: Stockholm University.
- Okoth Okombo, Duncan. 1997. A functional grammar of Dholuo (Nilo-Saharan 12). Cologne: Rüdiger Köppe.
- Paducheva, Elena V. 1998. On non-compatibility of partitive and imperfective in Russian. Theoretical Linguistics 24(1). 73–82.
- Pakendorf, Brigitte. 2010. Contact and Siberian languages. In Raymond Hickey (ed.), *The handbook of language contact*, 714–737. Malden, MA: Wiley-Blackwell.
- Park, Indrek. 2012. A grammar of Hidatsa. Bloomington: Indiana University dissertation.
- Partee, Barbara H. 2008. Negation, intensionality, and aspect: Interaction with NP semantics. In Susan Rothstein (ed.), *Theoretical and crosslinguistic approaches to the semantics of aspect (Linguistics today 110)*, 291–317. Amsterdam & Philadelphia: John Benjamins. https://udrive.oit.umass.edu/partee/ParteeInPressNegIntensAsp.pdf.
- Paul, Ileana. 2000. Malagasy clause structure. Montreal, QC: McGill University dissertation. http://publish.uwo.ca/~ileana/papers/dissertation.pdf.
- Paykin, Katia. 2014. The Russian partitive and verbal aspect. In Silvia Luraghi & Tuomas Huumo (eds.), Partitive cases and related categories, 379–398. Berlin & Boston: De Gruyter Mouton.

- Pennacchietti, Fabrozio A. 1974. Appunti per una storia comparata des sistemi preposizionali semitici. AIUON 24. 161–208.
- Persohn, Bastian. 2017. The Verb in Nyakyusa: A focus on tense, aspect, and modality. Berlin: Language Science Press.
- Persohn, Bastian. 2017a. Locative and extra-locative clitics in Nyakyusa. *Africana Linguistica* 23. 151–165.
- Persohn, Bastian & Maud Devos. 2017. Post-final locatives in Bantu axes of variation and non-locative functions. *Africana Linquistica* 23. 3–46.
- Quine, Willard Van Orman. 1960. Word and object. Cambridge, MA: The MIT Press.
- Reed, Ann M. 1989. Discourse groups and semantic groups. In Joyce Powers (ed.), *Proceedings of the 5th Eastern States Conference on Linguistics (ESCOL'88)*, 416–427. Columbus, OH: Ohio State University.
- Sasse, Hans-Jürgen. 2002. Recent activity in the theory of aspect: Accomplishments, achievements, or just non-progressive state? *Linguistic Typology* 6. 199–271.
- Schneider, Cynthia. 2010. A grammar of Abma: A language of Pentecost island, Vanuatu. Canberra: Pacific Linguistics.
- Seiler, Walter. 1985. *Imonda, a Papuan language (Pacific Linguistics, Series B, 93*. Canberra: Canberra: Australian National University.
- Selkirk, Elizabeth. 1977. Some remarks on noun phrase structure. In Peter W. Culicover, Thomas Wasow & Akmajian Adrian (eds.), Formal syntax: Papers from the MSSB-UC Irvine conference on the formal syntax of natural language, Newport Beach, California, June 9–11, 1976, 285–316. New York: Academic Press.
- Seržant, Ilja A. 2012. Morphosyntactic properties of the partitive genitive in the subject position in Ancient Greek. *Indogermanische Forschungen* 117. 187–204.
- Seržant, Ilja A. 2014a. The independent partitive gentive in Lithuanian. In Axel Holvoet & Nau Nicole (eds.), *Grammatical relations and their non-canonical encoding in Baltic (Valency, Argument Realization and Grammatical Relations in Baltic 1*), 257–299. Amsterdam & Philadelphia: John Benjamins.
- Seržant, Ilja A. 2014b. The independent partitive genitive in North Russian. In Ilja A. Seržant & B. Wiemer (eds.), Contemporary approaches to dialectology: The area of North, Northwest Russian and Belarusian vernaculars (Slavica Bergensia 13), 270–329. Bergen: John Grieg AS.
- Seržant, Ilja A. 2015a. Independent partitive as a Circum-Baltic isogloss. *Journal of Language Contact* 8, 341–418.
- Seržant, Ilja A. 2015b. An approach to syntactic reconstruction. In Vitti Carlotta (ed.), *Perspectives on historical syntax*, 117–154. Amsterdam & Philadelphia: John Benjamins.
- Seržant, Ilja A. 2020. Dataset for the paper "Typology of partitives", Linguistics [Data set]. Zenodo. https://doi.org/10.5281/zenodo.4277624.
- Seržant, Ilja A. submitted: Diachronic typology of partitives. In Giuliana Giusti & Petra Sleeman (eds.), *Partitive determiners, partitive pronouns and partitive case. (Linguistische Arbeiten)*. Berlin: De Gruyter Mouton.
- Seržant, Ilja A., Katarzyna Maria Janic, Darja Dermaku & Oneg Ben Dror, forthc. The typology of coding patterns and frequency effects of antipassives. *Studies in Language*.
- Smeets, Ineke. 2008. A grammar of Mapuche (Mouton grammar library 41). Berlin & New York: Mouton de Gruyter.
- Smith, Carlotta. 1997. The parameter of aspect, 2nd edn. Berlin: Springer.

- Strobel, Thomas & Elvira Glaser. 2019. The rise and fall of partitive markers in some Germanic varieties. In Tabea Ihsane & Elisabeth Stark (eds.), *Bare nouns vs. 'partitive articles': Disentangling functions*, 17–53. Leiden: Brill.
- Tănase-Dogaru, Mihaela. 2017. Partitive constructions. In Martin Everaert & Henk C. van Riemsdijk (eds.), *The Wiley Blackwell companion to syntax*, 2nd edn, 1–30. Hoboken, NJ: John Wiley & Sons.
- Terrill, Angela. 2003. *A grammar of Lavukaleve (Mouton Grammar Library 30)*. Berlin & New York: Mouton de Gruyter.
- Townsend, Elaine M., D. Marlene Ballena & Kenneth E. Swift. 1988. *Morfología del caquinte* (arawak preandino (Serie Lingüística Peruana 25). Lima: Ministerio de Educación and Instituto Lingüístico de Verano.
- Tveite, Tor. 2004. *The case of the object in Livonian: A corpus based study*. Helsinki: University of Helsinki MA thesis.
- Ubrjatova, Elizaveta Ivanovna. 1985. *Jazyk noril'skich dolgan [The language of the Norilsk Dolgans]*. Novosibirsk: Nauka.
- Ubrjatova, E. I., L. N. Xaritonov & N. E. Petrov. 1982. *Grammatika sovremennogo jakutskogo literaturnogo jazyka: Fonetika i morfologija [The grammar of Modern Yakut: Phonetics and morphology]*. Moscow: Izdatel'stvo Nauka.
- Vari-Bogiri, Hannah. 2011. *Phonology and morpho-syntax of Raga*. Vanuatu. Suva: University of the South Pacific dissertation.
- Westerståhl, Dag. 1985. Determinders and context sets. In Alice ter Meulen & Johan van Benthem (eds.), *Generalized quantifiers in natural language*, 45–71. Dordrecht: Foris.
- Wierzbicka, Anna. 1967. On the semantics of the verbal aspect in Polish. In *To honor Roman Jakobson: Essays on the occasion of his seventieth birthday*, vol. 3, 2231–2249. Mouton: The Hague & Paris.
- Winand, Jean. 2015. The oblique expression of the object in Ancient Egyptian. In Eitan Grossman, Martin Haspelmath & Tonio Sebastian Richter (eds.), Egyptian-Coptic linguistics in typological perspective, 533–560. Berlin & Boston: De Gruyter Mouton.
- Witzlack-Makarevich, Alena & Ilja A. Seržant. 2018. Differential argument marking: Patterns of variation. In Ilja A. Seržant & Alena Witzlack-Makarevich (eds.), *Diachrony of differential argument marking (Studies in diversity linguistics)*, 1–40. Berlin: Language Science Press. http://langsci-press.org/.
- Zimmermann, Thomas Ede. 1993. On the proper treatment of opacity in certain verbs. *Natural Language Semantics* 1. 149–179.